Received 13 December 2022; revised 6 February 2023; accepted 13 March 2023. Date of publication 20 March 2023; date of current version 28 March 2023. Digital Object Identifier 10.1109/OJCOMS.2023.3257889

Toward Greener 5G and Beyond Radio Access Networks—A Survey

LINE M. P. LARSEN⁽⁾^{1,2} (Member, IEEE), HENRIK L. CHRISTIANSEN⁽⁾¹ (Member, IEEE), SARAH RUEPP² (Member, IEEE), AND MICHAEL S. BERGER² (Member, IEEE)

¹Department of Radio Access Networks, TDC Net, 0900 Copenhagen, Denmark ²Department of Electrical and Photonics Engineering, Technical University of Denmark, 2800 Lyngby, Denmark CORRESPONDING AUTHOR: L. M. P. LARSEN (e-mail: lil@tdcnet.dk) This work was supported by the Innovation Fund Denmark under Grant 1045-00047B.

ABSTRACT Mobile network traffic is increasing and so is the energy consumption. The Radio Access Network (RAN) part is responsible for the largest share of the mobile network energy consumption, and thus; an important consideration when expanding mobile networks to meet traffic demands. This work analyses how the energy consumption of future mobile networks can be minimised by using the right RAN architecture, share the network with other operators and implementing the most efficient energy minimising technologies in the RAN. It is explored how the different approaches can be realised in real life networks as well as the research state of the art is highlighted. Furthermore, this work provides an overview of future research directions for 6G energy saving potentials. Different energy saving contributions are evaluated by a common methodology for more realistic comparison, based on the potential energy saving of the overall mobile network overall energy consumption can be reduced by approximately 30%, corresponding to almost half of the RAN energy consumption. Following this, a set of guidelines towards an energy optimised mobile network is provided, proposing changes to be made initially and in the longer run for brownfield network operators as well as a target network for greenfield network operators.

INDEX TERMS Green RAN, C-RAN, VRAN, O-RAN, MOCN, MORAN, network slicing, 5G, 6G, AI.

I. INTRODUCTION

T HE INFORMATION and Communications Technology (ICT) sector accounts for more than 2% of the global greenhouse gas emissions [1]. As a comparison, this level of greenhouse gas emissions is the same as used by the whole aviation sector [2]. However, today people have the opportunity of joining meetings remotely, and especially after a global pandemic where everything was moved into cyberspace, flight traffic does not grow with the same speed as the Information and Communications Technology (ICT) traffic. The required capacity of ICT sector grows when the number of users increase, or if the same amount of users consume more capacity. An important part of the ICT sector and a digitisation gamechanger are the mobile networks. Due to their still increasing popularity, the mobile networks require continuous expansion to keep up with the demands required by a number of subscribers envisioned to surpass 9 billions within the next few years [3]. To serve this large and increasing amount of subscribers, the mobile networks must not only provide coverage everywhere but also capacity enough to satisfy user demands. Hence, it is crucial for network operators to avoid the energy consumption to grow as fast as the network expansions. Thus, future greenfield installations and brownfield network expansions or updates must ensure a minimised energy consumption.

A mobile network consists of three segments: Radio Access Network (RAN), Mobile Backhaul (MBH) and core network. The RAN consists of a number of base stations serving a number of cells, in which the users of the mobile network are connected to the current base station via radio waves. The RAN is connected to the core network via the MBH. The base stations are located at cell sites. At the



FIGURE 1. The elements of the Radio Acces Network (RAN). The figure illustrates how the antenna is connected to the Radio Unit (RU) via a coax cable, and the RU is connected to the Baseband Unit (BBU) inside the cabinet.

cell site, radio waves transmitted by a User Entity (UE) is received by an antenna and sent to a Radio Unit (RU) via a coaxial cable. The RU handles the Radio Frequency (RF) functional part of the conversion from radio waves to bits. The RU then transfers the user data over a fiber connection to the Baseband Unit (BBU) which handles the remaining Baseband (BB) functions. A part from the BBU and RU, a cell site consists of a cabinet, that holds the BBU and backup power sources, with power supply and cooling which can be from air conditioning. A cell site installation is illustrated in Fig. 1. The current installation type used by many operators today is the so-called distributed RAN architecture. This architecture refers to the RU and BBU both being installed at the cell site, close to the users. The distributed RAN architecture as well as the three network segments RAN, MBH and core network are illustrated in Fig. 2. The figure illustrates only one cell served by one RU and BBU. However, in reality many RUs and BBUs are installed at each cell site. The number of RUs and BBUs will depend on the number of frequency bands and sectors served at the current cell site.

User mobility makes an un-equal distribution of the traffic in the network and only 20% of the base stations in the RAN carry 80% of the total traffic [4]. This leaves many opportunities to improve the energy consumption of the RAN. This paper focuses on energy minimisation potential in Radio Access Technologies (RATs) from 4G and beyond. Hence, as RATs become more and more energy efficient when technology emerges, equipment from older generations of mobile networks become much more energy consuming in comparison. Therefore, it is necessary to decommission previous RATs and focus on new ones with a lower energy consumption and a much higher utilisation of resources.



FIGURE 2. The distributed RAN architecture where the base station functions are divided into a Radio Unit (RU) and a Baseband Unit (BBU). The BBU is connected to the core network via the Mobile Backhaul (MBH) network.

This survey investigates solutions for future mobile networks to minimise the energy consumption, with a focus on architectures, network sharing opportunities and the technological evolution, by creating a common foundation of the energy savings impact. In the remainder of this paper, Section II presents state of the art in academia and industry as well as our contributions. Section III models the energy consumption, which will be used for the remainder of the work as a common comparison template. Section IV presents architectural energy optimisation opportunities of the mobile network. Section V presents the technological evolution that will minimise energy consumption in the future and Section VI examines how the network can be shared between different operators. Section VII discusses the impact of the savings presented throughout this survey and provides a set of deployment guidelines and directions for brownfield and greenfield network operators. Finally the conclusion closes this survey.

Please be aware of attached list of acronyms, placed right after the conclusion and acknowledgement on page 25.

II. RELATED WORK

Several surveys have previously looked into minimising the energy consumption of mobile networks. The work in [5] presents suggestions to improve and monitor the energy efficiency in 5G and beyond mobile networks and highlights previous projects focusing on energy efficiency. In [6], the authors provide a review and analysis of channel capacity, offline and online transmission schemes, and power optimisation from an information theory perspective. The survey

Contribution areas of this work	Other green RAN surveys contribution areas
Timely – published in 2020 or later	[6], [8], [9], [12], [14], [16], [17], [18], [19], [20], [21], [22], [23], [24], [25]
Cloud-RAN	[5], [7], [8], [9], [10], [11], [12], [13], [14], [15], [17], [21], [22]
Open RAN	[20], [21], [22]
Virtual RAN	[5], [7], [8], [10] , [13], [18], [21]
Network sharing	
Network slicing	[7], [8], [13]
Sleep modes	[23], [24]
Power amplifier improvements	[24]
Artificial Intelligence	[5], [7], [8], [20], [23], [25], [26]
Renewable energy	[5], [22]
Energy harvesting	[6], [7], [12], [17], [26]
B5G and 6G energy saving directions	[27], [16], [17], [18], [19]
A critical review of litterature	
Energy savings overview and comparison	[24]
Overview of research projects	[5], [8], [21], [28]

TABLE 1. Summary of existing surveys contribution areas compared to the contribution areas of this current work.



FIGURE 3. Research projects involved in activities aiming to reduce mobile network energy consumption. The x axis shows the years the project have run or is expected to run, the y axis shows the area of contribution.

in [7] considers the interaction between energy harvesting technologies and software based optimisation. The work in [8] also investigates how softwarisation can improve the energy effectiveness as well as the potential of adaptive network sharing with a focus on Cloud RAN (C-RAN) architecture. Table 1 compares existing surveys in the area of energy efficient mobile networks and green RAN to the contribution areas of this current survey.

As illustrated in Table 1, the C-RAN mobile network architecture, which has for several years been a vital part of surveys on energy reductions in mobile networks, is widely examined for examples in recent work: [5], [7], [8], [9]. However, the work in [10], [11], [12], [13], [14] addresses the C-RAN architecture particularly. More specifically, authors in [10] survey the benefits proposed in the case of C-RAN, for example minimised interference, delay and improved energy consumption. The literature survey in [12] highlights benefits and limitations of C-RAN as well as discusses potential enhancement techniques. The survey in [13] compares different network architectures with various amounts of shared resources. In [15], different functional constellations of the C-RAN architecture are surveyed.

A. GREEN NETWORKING RESEARCH PROJECTS

During the past years, energy reduction in mobile networks has received tremendous focus from the industry. Selected research projects are listed in this section, their timeline and areas of contribution are illustrated in Fig. 3.

- SOOGREEN: The "Service-oriented optimization of Green mobile networks", SooGreen project, investigated how to reduce the energy consumption of services in light of the traffic evolution and exploit new network architectures [29]. Results include energy optimisations powered by smart grid [30], hybrid C-RAN [31], and base station On/Off switching proposals [32].
- CHARISMA: The "Converged Heterogeneous Advanced 5G Cloud-RAN Architecture for Intelligent and Secure Media Access", CHARISMA project, investigated an intelligent, hierarchical routing and para-virtualised architecture, uniting the concepts of devolved offload decisions with shortest path nearest to end-users and an end-to-end security service chain [33].
- iCIRRUS: The "intelligent Converged network consolIdating Radio and optical access aRound USer

equipment", iCIRRUS project, proposed an intelligent C-RAN solution bringing together optical fibre technology, low-cost but highly flexible Ethernet networking and wireless resource management [34]. Results showed resource allocation and energy minimisation in C-RAN [35], [36], [37], [38], [39] and energy efficient scheduling [40].

- 5G-TRIDENT: The "5G Technologies for Reconfigurable and efficient raDio accEss iNfrastrucTure and mobile devices", 5G-TRIDENT project, investigated design and implementation aspects of 5G, with special emphasis on reconfigurability as well as cost- and energy-efficiency enhancement [41].
- One 5G: The "E2E-aware Optimizations and advancements for the Network Edge of 5G New Radio", one 5G project, investigated 5G extensions and provided a larger study on massive MIMO (mMIMO) targeting an optimised antenna design and a reduction of complexity and energy consumption [42].
- MonB5G: The MonB5G project, develops a hierarchical, fault-tolerant, automated data driven network management system that is incorporating security and energy efficiency as key features, for orchestrating a massive number of parallel network slices and more diverse service types using zero touch adaptive networking [43]. Results include energy optimisations for 6G [44] and increased infrastructure resource efficiency [45].
- DAEMON: The "Network intelligence for aDAptive and sElf-Learning MObile Networks", DAEMON project, investigates the use of network intelligence to deliver extremely high performance, reduce the energy footprint and provide extremely reliable networks [46]. Results include automated energy savings [47], [48], [49].
- DEDICAT 6G: The "Dynamic coverage Extension and Distributed Intelligence for human Centric Applications with assured security, privacy, and Trust: from 5G to 6G", DEDICAT 6G project, develops a smart connectivity platform using artificial intelligence and blockchain techniques with the aims to improve resource utilisation, reduce latency, response time, and energy consumption as well as cost reductions and reinforcement of security, privacy, and trust [50].
- REINDEER: The "REsilient INteractive applications through hyper Diversity in Energy Efficient RadioWeaves", REINDEER project, develops a smart connectivity platform as energy-efficient, smart, scalable and secure connectivity infrastructure, and topologies for zero-outage and efficient and secure deployment, bringing the intelligence closer to the user [51]. The particular "RadioWeaves" distributed infrastructure used, has shown improvements in quality of service and energy efficiency, compared to conventional collocated Multiple Input Multiple Output (MIMO) systems [52]. Recent efforts in the RadioWeaves technology include the work in [53].
- VOLUME 4, 2023

• Al4Green: The "Artificial Intelligence for Green Networks", Al4Green project works toward comprehensive, sophisticated and energy-efficient algorithms to solve the challenges faced in network design, deployment, and management [54]. The main goal of the Al4Green project is to achieve an improvement of about 30-40% of the end-to-end energy efficiency compared to current networks [54]. Results in the area on energy reduction contributions include: [5], [55], [56].

B. STATUS IN INDUSTRIAL ALLIANCES

Greener mobile networks receive tremendous attention from the industry. GSM Association (GSMA) is engaging the mobile industry towards climate action, and provides several yearly analysis on how the mobile industry is progressing towards the ambition to be net zero [57]. The O-RAN alliance addressed a set of energy efficiency goals and targets to support sustainable Open RAN deployments in their technical priorities of release 2 [58]. Next Generation Mobile Networks (NGMN) released in 2021 a whitepaper concerning green future networks [4]. Furthermore, the Next G Alliance have established a working group, referred to as Green G, which was launched in 2021 [59]. 3rd Generation Partnership Project (3GPP) is also working actively towards reduced energy consumption in mobile networks and have released a technical specification for energy efficiency of 5G [60].

C. OUR CONTRIBUTIONS

This current work presents an overview of methods for minimising the energy consumption in mobile networks with three major focus areas: Network architectures, network sharing and technology evolution. Thus, this paper analyses and compares the potential of current and future energy consumption minimisation techniques, with a critical review of existing literature and presents a methodology for a common comparison of the energy saving potential in different solutions. Furthermore, this paper provides guidelines for deployment and expansions of future mobile networks in a more energy efficient manner. As illustrated in Table 1, this combination of energy reducing parameters has not been seen before, as well as the critical literature review and the guidelines add a so far unseen dimension to the paper. Hence, the contributions of this paper are not only an overview of current research directions and their impact but also an overview of how proposed changes can be realised in future mobile networks. Hence, our contributions include:

• A critical review and analysis of existing literature in the area of network architectures, network sharing opportunities and technological evolution. Proposing a common comparison platform, to evaluate the energy minimisation impact of different proposals. Furthermore the energy savings are examined in perspective of possible Customer Experience (CEX) impacts.



FIGURE 4. The energy consumption of the mobile network in the inner circle, showing 73% energy consumption in the RAN, coloured in blue. The share of the core network is coloured in orange, the share of datacentres is grey and the share of other operations is coloured in yellow. The middle circle shows the share of the RAN, divided out on base stations and infrastructure. The outer circle shows the base station units: Radio processing, air conditioning, Baseband (BB) processing, power supply and main control.

- Proposing changes to the current network operator business model. Examining different opportunities on how network sharing can be realised.
- Inclusion of various deployment methods and opportunities for the different architectures and technologies proposed.
- An overview of research directions toward a more energy efficient 6G.

III. MODELLING THE RAN ENERGY CONSUMPTION

The mobile network traffic is continuously increasing, requiring continuous capacity extensions and expansions of the network. While, mobile network traffic is growing tremendously, nobody wants the energy consumption to grow with the same pace. Thus, methods to reduce the energy consumption of the mobile network is a widely researched area. However, different works propose different energy saving potentials and have different ways of calculating and presenting them. Hence, until now it has been difficult to gain an overview of the contributions of different works, and this is what this model provides: A simple comparison of the impact of different proposals in the bigger perspective of the overall mobile network energy consumption.

A. RAN ENERGY CONSUMPTION

In order to minimise the energy consumption of the mobile network, a starting point is, to understand where in the network, the energy is consumed. Looking into the energy consumption of the mobile network, then the RAN consumes 73% of the energy, the core network consumes 13%, the datacentres consume 9% and other operations account for 5% of the energy consumption [4], [61]. Figure 4 illustrates where energy is consumed in the mobile network.

The RAN energy consumption includes base stations as well as all associated infrastructure such as inverters, rectifiers, repeaters and MBH transport [61]. The energy consumption of the base station consists of 40% air conditioning, 40% radio processing, 7% power, 6.5% baseband processing and 3.5% main control [4]. The work in [4] is not more specific in defining what the different terms cover, therefore; some assumptions had to be made for this current work:

• Radio processing: is assumed to cover the energy consumed by the RUs and the antennas, which may vary by the amount of functions installed in the RU and the size of the antenna system as well as whether the antennas are active or passive. However, the baseline assumed is todays widely used RUs including only RF processing [62] and an active antenna system [63].

- Baseband processing: is expected to be the lower layers of BB processing, ie. Media Access Control (MAC) [15] and Radio Link Control (RLC) [15].
- Main control: is expected to be the upper control layers of the BB processing, handling inter site communication, ie. Packet Data Convergence Protocol (PDCP) [15] and Radio Resource Control (RRC) [15].
- Power supply: is expected to be cooling in the cabinet and the AC/DC conversion [64].
- Air condition: represents a large number in this context, but in reality the use of air condition is depending on the climate, and it can vary through different seasons. Hence, air condition is merely used in warmer climates and in summer times. For instance in [65] the authors set the air conditioning to 23 degrees Celsius, while the outer temperature will vary during day and night times as well as over seasons.
- The survey in [4], does not address the infrastructure energy consumption, but adding up the energy consumption of the remaining RAN installations, leaves 3% for the infrastructure.
- It is expected that the numbers come from distributed RAN installations [66].

The mobile network consumes a large amount of energy, and as illustrated in Fig 4, the base stations in the RAN are responsible for the majority of the energy consumption. Hence the first step for reduced energy consumption is to understand where in the network, most energy is consumed - where will changes have the highest impact? Another approach is to investigate, how can things be done differently than today? Today's mobile networks - or at least the majority of them (examples of other installation types will be provided throughout this survey), are run by one single operator, leading to multiple networks covering the same piece of land. Furthermore, the vast majority of network operators use only one RAN vendor or a minor group of vendors, potentially leading to less innovative solutions and longer time to market. At the base station, the most energy consuming element is the RU, and when one single RU is required for each band at each sector, many RUs must be installed at one single cell site. The users of a mobile network must always have coverage and capacity enough to fulfill their needs, which is particularly important since mobile networks provide critical infrastructure to today's society. Thus, mobile networks must be "always on", and constantly ready to respond to a user's request. This leads to under-utilisation of the network resources, which is energy consumed without any user traffic in the network. Hence, today's mobile networks with distributed RAN installation leaves room for many possible improvements which will be discussed throughout Sections IV-VI.

B. ENERGY CONSUMPTION COMPARISON

The model used by this paper provides an overview of the energy consumption of a particular unit in the network compared to the contribution in a bigger perspective. Hence, it is



FIGURE 5. An illustration of the three measurable network levels: Network - the overall mobile network, RAN and unit - illustrated by the BBU and RU.

used for comparing the contribution of different works, but also for evaluating the energy consumption of specific elements to be used in the guidelines proposed in Section VII. The model concept is introduced in Eq. (1):

$$\varepsilon_Y^X = \varepsilon_Y^Z \cdot \varepsilon_Z^X \tag{1}$$

In Eq. (1), ε is the energy consumption in percent, then Y is the object under investigation and X is what the energy consumption is relative to. Z is the unit relating X and Y. Hence, calculating how much the energy consumption of the different base station elements takes up of the overall mobile network energy consumption is expressed in Eq. (2), where three levels are used, corresponding to X, Y and Z, these levels are illustrated in Fig. 5:

- Network: The overall mobile network
- RAN: consisting of base station and infrastructure
- Unit: Representing the different units in the RAN, ie. BB processing, power supply...

The percentile energy consumption achieved by different proposals of energy consumption reductions of different units, is calculated as:

$$\varepsilon_{Unit}^{Network} = \varepsilon_{Unit}^{RAN} \cdot \varepsilon_{RAN}^{Network} \tag{2}$$

Hence, for example if a proposal claims a 15% reduction in the radio processing energy consumption, it corresponds to a reduction of 4.4% of the overall mobile network energy consumption. Therefore, this model can be used to determine the impact, different energy reduction proposals will have on the overall mobile network energy consumption and thus; it is possible to compare different proposals. The energy consumption numbers provided in this section will be used in the remainder of this paper, to translate expected energy savings into the overall saving of the whole mobile network perspective. Finally the overall energy saving contributions are evaluated in Section VII.

IV. ENERGY EFFICIENT NETWORK ARCHITECTURES

Minimising the energy consumption is an agenda that has been researched widely within the field of mobile network architectures, revealing the benefits of shared processing and minimised amplifiers by the use of RUs. Hence, the potential energy savings when changing the RAN architecture is worth investigating. This section looks into the energy saving contributions enabled when changing the traditional distributed RAN architecture to virtual RAN (vRAN), C-RAN or open RAN. Three network architectures that can be combined or deployed independently of each other.

A. VIRTUAL RAN

A virtualisation of the RAN, defines the separation and decoupling of Hardware (HW) and Software (SW), and thus; enables the use of Commercial off the Shelf (COTS) HW in the RAN, as the network functions run in SW. This separation opens up for more specialised providers, which can potentially reduce the energy consumption of the RAN. Hence, using vRAN the baseband processing does not need to be a "one black box" solution, but network operators can choose the most energy efficient HW and the SW with the most energy optimising functions. vRAN is realised by a conversion of the network functions into Virtual Network Function (VNF)s which is an abstraction of network functions into SW. VNFs are SW applications that deliver network functions, deployed as virtual machines [67]. The infrastructure and provisioning of new VNFs are handled in the Management and Orchestration (MANO) framework [67]. The work in [68] reports on energy savings of 30% when VNFs are introduced in 5G. vRAN areas of potential reduced energy consumption counts:

- COTS hardware
- No location dependency
- Improved operational efficiency
- Power down un-used HW

The vRAN technique is an enabler for many energy saving opportunities. However, from an energy minimisation perspective, this work will focus only on the above mentioned benefits in the vRAN context. Hence, many opportunities described later in this paper will rely on virtualisation of the RAN functions.

COTS HW in an energy efficiency context, relates to the use of HW delivered by vendors who specialise in building the most energy efficient HW. Hence, dedicated HW is built to match its purpose and COTS HW is built to match many purposes. For example when multiple 5G workloads must be run simultaneously on the same HW multi-domain network functions, adds complexity [69]. Based on the use of COTS HW, Deutsche Telecom reports on up to 30% Central Processing Unit (CPU) power savings in their vRAN testlab environment based on realistic traffic profiles [70]. These savings are achieved using scalable processors and power consumption management. However, it is not clear what the baseline for these CPU power savings are. The NGMN report [4] proposes a 15% energy efficiency increase using the latest generation COTS systems, however; the source does not comment on how these power savings are achieved.

Contradictory, the whitepaper [69] by Zain Group, stc, e&, Mobily, du, Batelco and Omantel, claims that dedicated HW consumes less energy compared to general purpose COTS servers. No timely and well documented research to back up the claims about COTS HW energy efficiency has been found, hence; it is not assumed to bring any energy consumption improvements. Furthermore, it is expected that the impact in CEX from choosing COTS HW, can vary with different vendors and products.

No location dependency will benefit the energy consumption in the way that processing can, in principle run on any server in the world - if it was not limited by latency. Hence, the processing can be run on the server where the highest degree of resource utilisation is met. On the other hand, energy consumption can also be saved due to shorter paths, and thus less energy consumption in the MBH [68]. In this regard, the processing is moved as close to the user as possible.

Improved operational efficiency is achieved when the virtualisation platform works in high-performance modes to optimally utilise the CPU powers [68]. This, however; requires for the CPU capacity to be fully utilised in order to have any effect on the energy consumption.

HW can be powered down when not in use, since vRAN will enable the BB SW to run on selected HW, leading to every RU not being attached to one single BBU, but rather one RU sending data to the HW with the highest resource utilisation. Hence, when HW is not in use, it can be periodically powered off or used for other purposes. Furthermore, HW sleep states can be used in low activity periods, to gradually shut down the HW [71]. The work in [72] achieves an average energy saving of 34% by optimising the location and utilisation of virtual machines running VNFs.

Energy saving potential: vRAN technology is the enabler of many power saving technologies, which will be mentioned later in this analysis, as well as the enabler of many features that will improve the CEX. On the other hand, the use of COTS HW can potentially lead to a decrease in CEX, this is yet to be investigated.

B. CLOUD RAN

Its is hard to say energy efficient network design without mentioning C-RAN. The initial approach of C-RAN was to minimise the RU and move it to the antenna tower for the amplifier to become as small as possible, leaving as much processing as possible to be shared in a centralised BBU datacenter. The C-RAN centralisation of processing functions introduces the very capacity demanding and latency strict fronthaul network, which is required to carry raw Inphase Quadrature (IQ) symbols in the Common Public Radio Interface (CPRI) protocol. To reduce these large fronthaul requirements, more functions were added to the RU. The exact division of functions left local in the RU and functions to be centralised is referred to as the functional split [15]. Hence, the trend is pointing towards more functions in the



FIGURE 6. The evolved Cloud-RAN architecture where the radio functions are found in the Radio Unit (RU) and the baseband functions are divided into the Distributed Unit (DU) and Centralised Unit (CU), connected by the crosshaul network. The crosshaul network describes the fronthaul, midhaul and backhaul networks. The DU and CU can be placed at the same or different locations, illustrated by the blue arrow. The baseband processing functions are virtualised.

RU and less shared functions, which is not minimising the BB processing energy consumption.

The work in [73] and [74] show how the energy consumption of the fronthaul network is reduced when more functions are left in the RU. Hence, the energy consumption of C-RAN is a trade-off between shared processing and transport network. In [75], the correlation between energy consumption in RU, fronthaul and BBU is also explored showing how energy consumption is decreasing when the delay on task execution and data transmission is increasing. The energy consumption of the fronthaul network is dependent on a large amount of parameters depending on technology, for example required amount of Small Formfactor Pluggable (SFP)s as well as the number of hops and amplifiers along the path [74]. Hence, the trade-off between saving processing energy consumption and saving fronthaul energy consumption leaves room for further investigations. The work in [76] proposes a resource allocation scheme to explore the trade-off between fronthaul load and functional split.

Recent C-RAN architectures used by O-RAN Alliance [77], NGMN [78] and IEEE 1914 [79] divides the baseband processing into two units: Distributed Unit (DU) and Centralised Unit (CU). This opens up the opportunities for new mobile network architectures - whose energy efficiency is yet to be investigated. The network between the RU and DU is referred to as the fronthaul network, and the network between the DU and CU is referred to as the midhaul network. The fronthaul, midhaul and MBH networks form together the crosshaul network. The crosshaul network, as well as the RU, DU and CU constellation are illustrated in Fig. 6. The functional split separating the RU and DU functions are referred to as the Low Layer Split (LLS), and the functional split separating the DU and CU are referred to as the High Layer Split (HLS). HLS is selected as separating the RLC and PDCP functions [80], this is illustrated in Fig. 7. The LLS has not yet been determined and different approaches are proposed by different organs and alliances, examples can



FIGURE 7. The functional division of High Layer Split (HLS) and proposals for Low Layer Split (LLS), where option 8 is the original C-RAN functional split. Option 7-2 is very interesting seen from an energy optimisation point of view since it improves the resource utilisation in the fronthaul network.

be seen in Fig. 7. From an energy optimisation point of view, functional split option 7-2 is very interesting, since it has the resource element mapper inside the RU, enabling a variable load on the fronthaul network [15], and thus; potentially higher utilisation of the fronthaul network resources. The fronthaul and midhaul networks are limited by different latency parameters. Therefore, the fronthaul transport must be $< 250 \ \mu s$ [80] and the midhaul transport must be $< 10 \ ms$ [80]. This can be converted into a maximum transmission distance using the fiber propagation delay, which is here assumed to be $10 \ \mu s/km$ [81]. Hence, seen from an energy reduction perspective, where as much shared processing as possible is wanted, we have two scenarios:

- 1) To place all DUs within a range of < 25 km transport in one datacenter and all CUs within a range of < 1000 km in another datacenter.
- 2) To place all DUs and CUs within a range of < 25 km transport in one datacenter.

The energy consumption impact of choosing between either of the two above scenarios is still an uninvestigated area. However, deployment scenarios where DUs or both DUs and CUs are located at the cell sites are seen as less energy optimised, since less or no functions are shared between sites.

To achieve most benefits from the baseband pooling in C-RAN, the baseband functions must be virtualised, which is enabled by vRAN. The energy minimising improvements benefitting from the virtualised and shared baseband processing enabled by C-RAN are listed below, and will be further elaborated on afterwards:

- Coordinated transmission
- Load consolidation
- Reduced Processing
- Fronthaul optimisation
- RU optimisation

Coordinated transmission implicates the RUs transmitting to the same user, improving network throughput and reducing interference. The work in [82] shows how energy efficiency, increasing the number of bits per joule; is improved by 51% using their proposed coordination scheme. However, this term does only define the improved utilisation of the network, and not how much energy consumption can be reduced. Another benefit of centralising the baseband processing is the support of Coordinated Multipoint (CoMP), which minimises interference and improves system performance [15]. As CoMP improves the Signal to Noise Ratio (SNR), it will also lower the required transmission power [8], and this will reduce the requirements of the power amplifier. The implications and benefits of CoMP and C-RAN are surveyed in [83]. The improvement from using coordinated transmission technologies is estimated to reduce the energy consumption of the radio processing by 5% and is expected to increase the CEX.

Load consolidation is by means the utilisation of having baseband processing from multiple cell sites gathered in one datacenter. Thus, this leads to the number of processing resources being minimised because resources can be assigned where they are needed, and de-activated when not needed. The work in [84] proposes two state transition strategies and shows energy saving efficiencies up to 74%. However, the energy saving efficiency is derived by dividing the maximum power of the BBU pool with the activation of all BBUs in the pool [84]. Hence, the energy saving efficiency cannot be compared to actual energy savings of using the two proposed strategies. The work in [85] illustrates the energy improvements of cloud processing compared to cell site processing in a distributed RAN architecture. This work measures the energy efficiency in bits/joule and with respect to both the channel correlation coefficient and the threshold [85]. In [86], a dynamic load consolidation algorithm is proposed which saves 21% power for the BB processing. This translates into a saving of 1% of the overall mobile network energy consumption. Authors in [87] investigate the optimisation problem of resource allocation and power control achieving an energy efficiency improvement of 57%. In [88], energy consumption savings up to 93% are proposed for a virtualised cloud-RAN. However, authors in [88] claim the energy consumption savings will decrease when adding more servers to the pool, that does not sound like good load consolidation, hence; it is expected that the energy consumption savings will increase when more servers are added to the pool. The improvement from using load consolidation techniques is estimated to reduce the BB processing and main control energy consumption by 20% and is not expected to have an impact on the CEX.

Reduced processing can be achieved in various ways. The work in [89] uses parallel processing, where each job runs on a single core and only one at the time, is used to minimise the runtime of the BB processing. In regard to energy efficiency, a reduced runtime will make room for more processing at

the current server, and thus; higher resource utilisation. The work in [90] proposes an association scheme for reduced overhead between RU and BBU. Results show a 20% reduction in power consumption of the BB processing [90], this corresponds to almost 1% of the overall mobile network energy consumption. The improvement from using reduced processing techniques is estimated to reduce the BB processing energy consumption by 20% and is not expected to have an impact on the CEX.

Fronthaul optimisation provides energy consumption reductions when fronthaul resources are better utilised or the RU has more functions, reducing the fronthaul transmission rate. The work in [91] develops a successive convex approximation algorithm, which optimises the MIMO transmit powers, while lowering the fronthaul capacity. Results in [91] show energy efficiency improvements of 49% using the power based optimisation. It is not clearly stated in which part of the system the energy efficiency improvements are achieved, but it is implicated that it is in the fronthaul transmission part. However, the energy efficiency does not translate into energy consumption and thus the savings can not be related to overall energy reductions. The work in [23] states the huge impact, the functional split has on the energy consumption of the mobile network. Hence, using functional split 6, the fronthaul energy consumption does only account for around 2% of the overall mobile network energy consumption, while using functional split options 7 and 8 it accounts for 30% and 60%, respectively [23]. The huge energy consumption in the fronthaul network compared to different functional splits are investigated in [73], showing large differences in the fronthaul energy consumption for different functional splits. Hence, these results show how the introduction of C-RAN and fronthaul networks can radically increase the energy consumption of the mobile network, and thus; the great importance of choosing the right functional split. Thus, the trade-off between energy consumption in the fronthaul network and energy savings related to shared BB processing is a topic that leaves room for further investigations.

RU optimisation is important, since the RU is the most energy consuming part of the mobile network. Originally C-RAN greatly improved the energy consumption by dividing base station functions into the RU and BBU, minimising the amplifier. Newer releases of the RU actually merge the RU and antenna into one unit, minimising the amplifier as much as possible [92], referred to as an antenna unit. It is claimed that the introduction of the antenna unit can decrease the energy consumption by up to 43% [93]. A 43% reduction of antenna unit energy consumption translates into a saving of 28% of the overall mobile network energy consumption. However, the antenna unit is developed for large antenna arrays and thus; it does not directly translate into RU power reductions. The improvement from minimising the power amplifier in the RU is estimated to reduce the radio processing energy consumption by 20% and is not expected to have an impact on the CEX.

Energy saving potential: According to the original thoughts of C-RAN provided in 2010 by IBM [94] and in 2011 by China Mobile [95], then the pooling of BB processing will greatly reduce air conditioning since it is not required on-site anymore. Thus, saving energy by one central air conditioning system seems like a valid way to save at least half of the energy consumed by air conditioning. A 50% decrease in air conditioning energy consumption corresponds to more than 14% of the mobile network energy consumption. However, it is noted that not all areas use the same amount of air condition due to different climates and northern countries does not use air condition at all, they use ventilation installed in the cabinet, whose energy consumption will not be improved when introducing C-RAN. C-RAN is by this work estimated to reduce energy consumption of the power supply by 10% due to reduced on-site footprint and centralised processing. In summary of this subsection, then C-RAN will introduce a 40% reduction in BB processing, a 20% reduction in main control energy consumption and a 25% decrease in radio processing energy consumption. Furthermore, C-RAN is not only expected to decrease power consumption but also a slight increase in the CEX is expected due to the higher degree of transmission coordination. However, this is at the cost of additional user latency [96].

C. OPEN RAN

The emerging open RAN technology opens up the interfaces in the mobile network. More specifically the fronthaul interfaces between RU and DU as well as the Xn interface between CUs receive special attention. Opening these network interfaces will open up the opportunities for a multivendor network with highly specialised suppliers, where the network operators can mix and match equipment delivered by different vendors. From an energy efficiency perspective, then this will provide network operators the opportunity to select network equipment based on its energy consumption. NGMN highlights, in a recent whitepaper [4], the importance of not only improving the major parts of the equipment but to take minor subparts and improve their energy consumption. This is exemplified by the RU where they state that not only the power amplifier's energy consumption must be reduced, but also the consumption of small signal and digital intermediate frequency modules [4]. This follows the key reasoning of open RAN where subparts are improved by highly specialised vendors.

O-RAN Alliance is a major driver for open RAN specifications, and their specifications merges the concepts of C-RAN, vRAN and open RAN. This is illustrated in Fig. 8. The proposed O-RAN system is built upon the Network Function Virtualisation (NFV)-MANO reference architecture proposed by European Telecommunications Standards Institute (ETSI) [97]. Hence, the SW is separated from the HW and with different layers of control systems on top. This opens up for even more individually specialised components, and thus also opportunities for energy optimisation. The



FIGURE 8. The O-RAN ecosystem embedding concepts of open RAN, Cloud-Radio Access Network (C-RAN) and virtual RAN (vRAN).

O-RAN contributors in the OpenRAN group, part of Telecom Infra Project (TIP), have in [58] released a set of technical priorities with focus on energy efficiency that includes Key Performance Indicator (KPI)s and features to improve energy efficiency. These features include radio features and power management at HW and cloud SW level [58]. In a laboratory setting, researchers from a number of companies achieved power consumption reductions of the O-RAN infrastructure by 9% in a peak traffic scenario and 12% during a low traffic scenario [98]. The source [98] does, however; not write anything about which components of the O-RAN infrastructure receives the power reduction or what the power reduction is compared to, to achieve given reductions. The O-RAN specifications provide the following power saving features:

- COTS HW
- Fronthaul
- Dynamic DU selection
- Intelligent control

COTS HW was already mentioned under vRAN.

Fronthaul energy reductions are achieved by lower fronthaul requirements, due to using functional split option 7-2x. The two categories A and B of functional split option 7-2x are illustrated in Fig. 7. O-RAN specifications promise significant lower fronthaul requirements compared to the original split 8 functional division carrying CPRI samples [99]. Further, the used transmission bandwidth grows proportional to the user bandwidth [99]. The DUs are not dedicated resources and can serve more than one RU [99]. However, O-RAN is not locked to one network architecture and thus the impact of fronthaul network energy consumption reductions will rely on the length of the fronthaul. Furthermore, functional split option 7-2x does not have the resource element mapper installed in the RU, which makes the fronthaul load constant.

Dynamic DU selection is a concept evolving in the O-RAN specifications, but based on the concept of no location dependency. The work in [100] is based on the fact that O-RAN

DU and CU processing pools can be located in different geographical places and have limited capacity, affecting the energy consumption and the performance of networks. Hence, authors in [100] propose a joint optimisation solution. When optimising the propagation delay between RUs and DUs, authors achieve at 40% improvement in energy consumption [100]. However, the authors do not state in which network element the energy reduction is received, and thus; it is difficult to compare the reduction to the bigger picture. In [101], the opportunity of executing DU and CU processing in different locations is explored, under the goal of choosing a proper location for running VNFs with respect to propagation delay and computational capacity. It is shown that by dynamically transferring the load to the most suitable DU, energy consumption is decreased by up to 50% [101]. The authors do not specifically state in which equipment the energy consumption is reduced, but it is assumed to be a reduction in the BB processing. A 50% reduction in the BB processing energy consumption corresponds to saving 2.4% of the energy consumption in the mobile network. The improvement from using dynamic DU selection techniques is estimated to reduce the BB processing energy consumption by 20%. However, it eliminates the benefits achieved by shared processing and load consolidation in C-RAN. Dynamic DU selection can potentially impact the CEX positively since the processing can be handled with the least amount of latency, if that is the criteria. On the other hand, it can also have a negative impact in case other criteria must be met.

Intelligent control is natively implemented in the O-RAN specifications inside the Radio Intelligent Controller (RIC)s which are deployed in near real time and non real time. RICs bring smart decision making into the RAN using Artificial Intelligence (AI) and Machine Learning (ML) frameworks. The work in [102] investigates resource allocation in the RIC using federated learning, in a network sliced environment where traffic must be forecast on a per slice level. Furthermore, the work in [103] surveys deep learning-based work for 5G and the integration to AI-enabled O-RAN architecture.

Energy saving potential: In general the open RAN concept itself does not add any other energy saving contributions than the potential improved energy consumption when choosing more specialised suppliers. However, in this work the energy savings enabled by the O-RAN specifications are also considered part of open RAN energy savings. The dynamic DU selection energy consumption reductions relate to 20% saving of the baseband processing energy consumption. However, the concept of dynamic DU selection is contradictory with the C-RAN shared DU processing. In [103], the three technologies of C-RAN, vRAN and O-RAN are compared and the authors claim that the centralisation of network functions in C-RAN causes the RAN energy consumption to decrease, and a virtualisation will reduce the energy consumption further. This is their claim to categorise C-RAN energy consumption as medium and that of O-RAN and

vRAN is low. The CEX impact of introducing open RAN is yet an uninvestigated area. However, the native intelligence of the O-RAN systems will definitely bring improvements to the CEX and potentially also to the energy consumption.

V. TECHNOLOGY EVOLUTION

New technologies keep improving the energy efficiency compared to older generations, as new methods for building, running and maintaining the network are developed. Thus, this section highlights improvements occurring when the current 4G or 5G network is optimised, when implementing 5G and outline what is expected when going from 5G to 6G.

A. TECHNOLOGY IMPROVEMENTS

With the emerging 5G technology, the mobile network is not only specialised for specific use cases, it is also optimised in terms of utilising the existing resources. The introduction of 5G is an on-going process, some operators have not started and some operators are in the long lasting implementation phase. However, 5G will bring new opportunities in regards to energy saving potential, as well as the existing 4G network can be optimised in many areas. Existing technology improvements that will result in energy consumption reductions are listed below:

- Power amplifier improvements
- Spectral efficiency
- Reduced signalling
- Sleep modes
- Network virtualisation
- Artificial Intelligence

Power amplifier improvements play a significant role in reducing the energy consumption of mobile networks and can be carried out in numerous ways in 5G. One solution is to use high-efficient semiconductor material Gallium Nitride (GaN) for maximising the power density. In [104] GaN is used to manufacture an RF power amplifier which increases the efficiency by 40%. Another solution is to use the same amplifier to support multiple bands. Multi-band Radio will reduce the energy consumption of the RU since the power amplifier plays a key role in the energy consumption of a RU. Thus, reducing the energy consumption of the power amplifier will have a large impact on the overall energy consumption of the mobile network. In multi-band radio, wideband technology is used where multiple bands can be served by the same amplifier, reducing the number of total amplifiers [105]. Ericsson claims that their multi-band radios have increased energy efficiency by up to 20% compared to the earlier generation [93]. Furthermore, in [106] a computer model to study power amplifier performance was proposed, which can be used to optimise the utilisation of the amplifier. Another way to reduce power amplifier energy consumption is to use MIMO power control. MIMO power control ensures how the user is always served by the base station with the best coverage conditions. This is because, if the coverage is decreased, then the UEs must use a higher transmission

power and when the UE is using resources at multiple base stations, the network will consume more energy. In this work, the improvements from using multi-band radios and GaN amplifiers are estimated to reduce the energy consumption of the radio processing by 20%. The use of GaN power amplifiers is envisioned to improve the CEX slightly where multi-band RUs can have a potential negative effect on the CEX in case of traffic congestion.

Increasing the spectral efficiency will make use of the spectrum more efficiently, and make the network more energy efficient as it requires less power to send the same amount of information. 5G includes many techniques for improving the spectral efficiency; filtered Orthogonal Frequency Division Multiplexing (OFDM), polar codes, Sparse Code Multiple Access (SCMA) technologies and mMIMO. mMIMO is a concept that significantly improves multiplexing and array gain of 5G transmission systems, since it is exploiting a large number of base station antennas for serving many users with the same time and frequency resources [107]. mMIMO systems with a large number of antennas installed, enable simpler precoding algorithms, signal detection and transmission at the base station, contributing to the reduction of the dissipated energy required for signal processing computations [107]. Furthermore, in mMIMO systems power amplifier losses during operations will be reduced which can bring significant power savings [107]. On the other hand, mMIMO increases the number of energy consuming Analogue-to-Digital Converter (ADC)s with corresponding digital circuitry and power amplifiers [107]. One of the 5G enablers, milimeter waves (mmwaves), consume much energy due to their complexity and because the technology requires analog precoders increasing the need for energy consuming signal mixers and ADCs [108]. Thus, hybrid analog-digital precoders are considered and mMIMO systems are used to reduce complexity and energy consumption [108]. The work in [109] investigates a mMIMO system with multiple arrays which must be located optimally to achieve most energy savings.

When *Reducing signalling*, less bits need to be transferred around in the network, leading to less operations and thus also less energy consumption. By introducing 5G, initiatives to simplify the signalling are already part of the network by use of the synchronisation signal block. Using the synchronisation signal block, continuous signals from 4G are in 5G broadcasted with a varying periodicity [4]. Furthermore, 5G allows for multiple granularities of time in-between signals, improving the resource utilisation by making room for sleep periods [4]. Hence, control signals are not consistently transmitted in every radio frame, but on demand and more sparsely, based on traffic requirements [23].

Sleep modes were developed as a way to get rid of underutilised resources, as mobile networks are deployed to meet peak hour demands. Hence, when resources are not in use, they are simply switched off. The lean carrier design of 5G, enables shut down of different base station components, or just longer periods between signals. Sleep modes come



FIGURE 9. The sleep modes: Symbol shutdown, MIMO sleep mode/muting and complete carrier shutdown. The figure illustrates how using symbol shutdown, the amplifier is shutdown which implicates a minor latency. Using MIMO sleep mode and MIMO muting, some of the antenna array is shutdown which implicates lower capacity. And using complete carrier shutdown, the whole cell shuts down which implicates a larger delay before it is in normal operation again.

in five different levels: Symbol shutdown, traffic shaping, MIMO sleep mode, mMIMO muting and complete carrier shutdown.

1) Symbol shutdown: Describes a periodical deactivation of RU power amplifier [4] the concept is also known as micro sleep [110]. Symbol shut down is foreseen to reduce power consumption by up to 30% [4]. The less required always-active signalling in 5G, described under "reduced signalling", allow for longer microsleep periods of up to up to 160ms in the presence of bursty traffic [23]. However, for a coverage providing 5G carrier the default periodicity is 20 ms [111]. This translates into an approximately 100 times longer silent period compared to a LTE carrier [111]. However, the price to pay is reduced CEX since a slight latency is added to the transmission when the power amplifier is activated again. The symbol shutdown sleep mode and its minor latency implication is illustrated in Fig. 9.



FIGURE 10. The 5G frame structure where traffic has been shaped to only transmit in certain subframes, in order to improve the outcome of the symbol shutdown sleep mode. Hence, the RU power amplifier can be deactivated for the duration of the empty subframes.

- 2) Traffic shaping: In 5G it is possible to create empty symbols, slots and subframes, by smart scheduling decisions [111]. Hence, traffic can be shaped for longer periods of symbol shutdown. Such techniques can be used to allow deeper sleep modes or improve the efficiency of the symbol shutdown [23]. However, these traffic shaping techniques comes at the cost of a minor latency. For example the Low Energy Scheduler Solution (LESS) [110] reschedules downlink transmissions for non-critical data [110]. This solution improves the symbol shut down efficiency because more timeslots are emptied and can trigger symbol shut down [110]. In [110] the combination of symbol shut down and LESS reduces the RU energy consumption by 15%. This corresponds to a 4% reduction of the overall mobile network energy consumption. Traffic shaping is illustrated in Fig. 10, where data is only transmitted in selected subframes, and the RU power amplifier can then be deactivated for the duration of the empty subframes.
- 3) MIMO sleep mode: Is the deactivation of a number of antenna branches in low traffic periods [110]. For example in the case of 8x8 MIMO or 4x4 MIMO the RU can be reconfigured to 2x2 MIMO mode in low traffic periods. Hence, the capacity of the cell is reduced in periods. The MIMO sleep mode and its reduced capacity implication is illustrated in Fig. 9.
- 4) mMIMO muting: Shutdown of channels in a low traffic period, results in reduced antenna gain leading to coverage loss [4], furthermore, the capacity of the cell is reduced in the muting period. Channel shutdown using mMIMO muting is foreseen to reduce power consumption by up to 20% [4], however; it is not stated in which network components the energy consumption is reduced. It is expected to be in the radio unit.

MIMO muting and its reduced capacity implication is illustrated in Fig. 9.

5) Complete carrier shutdown: In this sleep mode, capacity layer cells are shut down. This can be done down to only the coverage layer cells retained. Compared to symbol shutdown, then this approach has less gain and might have a higher impact on the CEX [4]. Hence, carrier shutdown deactivates HW components for longer periods of time, scaling from minutes to hours and thus; adapts to long term variations of a cell [23]. Furthermore, it takes time to power on the whole system again which will implicate latency. Therefore, carrier shutdown is seen as a key proposal in low traffic periods, for example at night times. Deeper sleep modes are referred to as Advanced Sleep Mode (ASM), defined in [112] as a progressive shutdown of the base station depending on the activation and the deactivation times of the different components. The complete carrier shutdown sleep mode and its latency implication is illustrated in Fig. 9.

In this work, introduction of various sleep-modes are estimated to reduce the energy consumption by 8% of the radio processing using symbol shutdown and traffic shaping, by 2% of the radio processing using MIMO sleep mode and mMIMO muting respectively. Furthermore, the energy consumption of the whole base station is estimated to be reduced by 2% using complete carrier shutdown sleep mode. However, for coverage layer cells, only MIMO sleep is seen as an option, where capacity layer cells can use a combination of symbol shutdown, traffic shaping and carrier shutdown. Hence, the total potential of sleep modes are estimated to be 12% of the radio processing and 2% of the remaining base station units. Sleep modes come at the cost of decreased CEX, since the users will experience delays and/or decreased throughput.

Network virtualisation is continuously used in future generations of mobile networks. Cloud-Native Network Function (CNF)s is the successor of VNFs by using containers rather than virtual machines [67]. Containers pack the SW with all of the files necessary to run it; while sharing access to the operating system and other server resources, where CNFs are designed and implemented to run inside containers [67]. Hence, containers do not package anything bigger than an app and all the files necessary to run it. Thus, containers are often used to package single functions that perform specific tasks. CNFs are decomposed into microservices, which allows multiple versions during updates, and using available platform services [67]. Thus, unlike virtual machine migration, used in VNFs, where the CPU state, memory content, and content storage are considered during the migration process, then the migration of containers is primarily concerned with memory content [113]. Another benefit of using containers is that the use of virtualisation layers, like virtual machines in VNF to virtualise network functions, introduces fluctuations in processing time. These fluctuations makes it challenging to maintain connectivity because of the frequently missing real-time requirements and thus; this problem has been overcome in CNF [113]. The work in [113] studies live migration of network functions where containerised functions are migrated between two hosts, while keeping the users' connection active. Results in [113] show that the RAN could continue operations and remain synchronised with the rest of the network despite downtime periods of up to 8 ms, which is effectively the period of a single Hybrid Automatic Repeat Request (HARQ) Acknowledgement (ACK). The improvement from using network virtualisation technologies is estimated to reduce the energy consumption at the same level as for vRAN described in Section IV.

Artificial Intelligence is an area of computer science focusing on creating machines that can engage on behaviors that humans consider intelligent [111]. Hence, in terms of networking AI can be used to compute decisions that were before made by humans, based on predictions and real time measurements. Thus, AI is made up of three principal branches; big data, automation and artificial intelligence [111]. In this regard, big data uses analysis of large datasets, automation uses pre-programmed rules and artificial intelligence applies cognitive functions to machines [111]. AI is a key enabler of optimisations in all aspects paving its way into 5G, and also in terms of energy efficiency. However, the complexity of AI increases when entering wireless communication channels, such as mobile networks, since the conditions of the channel will vary by time. AI based energy saving use cases for 5G include:

- Traffic load forecast [111]
- Service awareness [111]
- Threshold optimisation [111]
- Beam pattern optimisation [111]
- Energy harvesting optimisation [114]
- Optimise the use of renewable resources [114]
- Estimation of the direction of arrival [115]
- Signal propagation prediction models [116]
- Energy optimisation [117]

AI comes at the expense of both a very long exploration phase as well as large requirements to computing powers and storage [23]. Furthermore, the continuous adapting of an optimal policy to variations of network settings is a challenge, especially when it is derived from and for a limited set of specific system configurations [23]. The introduction of AI is not envisioned to bring any energy savings itself, it is merely the combination of AI, network virtualisation and sleep modes or improved spectral efficiency that will improve energy consumption. Several solutions for systems using AI to reduce energy consumption in mobile networks are already on the market today, examples count: Nokia [118], Huawei [119], ZTE [120], Samsung [121] and Ericsson [122]. Where multiple solutions report on power savings up to 20% energy consumption [118], [120].

A subfield of AI is ML, where the machines learn based on the training they receive. A subfield of ML is Deep Learning (DL), where the machines learn to make decisions



FIGURE 11. An illustration of the correlation between deep learning being a subfield of machine learning, being a subfield of artificial intelligence.

approaching a cognitive way of thinking. Further description of AI, ML and DL algorithms for communication can be found in [114], [123]. Figure 11 illustrates the correlation between AI, ML and DL. On a rough basis the three concepts can be defined as:

- AI: The machine responds to a set of rules
- ML: The machine can via learning adapt with minimal human interference [123]
- DL: The machine uses artificial neural networks to mimic the learning process of the human brain [123]

Below subsections focus on work done in the area of ML and DL.

Machine learning is a subfield of AI, which is widely used within mobile network predictions. ML enables machines to perform tasks by learning from data instead of running a static computer program [124]. Hence, ML uses statistical techniques to perform specific tasks, often requiring a smaller amount of data and with a lower time for application training [111]. ML tools can be divided into three types of learning: Supervised, unsupervised and reinforcement. The latter being learning through feedback. ML algorithms can provide both traffic predictions and energy predictions to enable utilisation of renewable energy sources [4]. ML can also be utilised to decode and encode transmissions, which will improve spectral efficiency by up to 20% [125]. 5G enhancements facilitates the usage of ML to derive optimal and network-wide energy efficient operation policies through centralised data gathering and processing [23]. Hence, the current 3GPP NR architecture introduces the Network Data Analytics Function (NWDAF) and the Management Data Analytics Function (MDAF) [23]. Within reinforcement learning, the Q-model is used in [112], [126], [127] and [128]. In [128], the authors utilise the Q-model to adapt a base station's resources according to the traffic demand, by optimising the use of sleep modes. Results in [128] reports on up to 16% energy consumption savings. Authors in [127] use a framework for effective strategy selection and optimisation policies and obtains up to 20%

energy savings in a high load scenario. The work in [126] utilises the Q-model to optimise the trade-off between QoS and energy efficiency. In [112] different cell loads are investigated using the Q-model, where for very low loads up to 55% energy savings can be achieved and 10% with a moderate load. Recent work in [129] proposes a ML mechanism that provides new sleep mode functionalities, where the greedy algorithm can save up to almost 80% of the energy. The work in [112], [128] and [129] do, however; not explain where in the network, the energy savings are achieved.

Deep learning is a subfield of ML, inspired of the human brain to solve problems that ML can not handle. Hence, the difference between DL and ML is bound in the learning complexity, and while both are AI, then AI is not DL nor ML [123]. DL uses artificial neural networks which require a more substantial amount of data to train models, allowing to process unlabeled data and solve end-to-end problems [111]. Hence, it utilises a set of tools for processing raw data. When large data sets are available, DL is known to outperform other ML techniques, since it is able to operate in a fully data-driven fashion [124] and is advantageous in high-dimensional data processing. The work in [130] uses a DL approach to optimise cell sleeping and beamforming, showing great improvements in performance. The authors in [131] utilises DL tools to subchannel assignment and power allocation strategies to optimise energy efficiency while maintaining bitrate conditions for the users. Also in [132], DL is used to evaluate the quality of candidate radio bands based on the spatial and spectral correlation between radio frequency signals. In [124], the authors show how Artificial Neural Networks (ANN) can be used to implement real-time resource allocation policies that are too complex for real-time by conventional theoretical optimisation approaches. Hence, the ANN tool is used to maximise the energy efficiency in real-time. The work in [133] uses deep reinforcement learning, which is a DL method to process high-dimensional input data and learn actions interactively from the environment, to achieve energy savings up to 27%. In [133] deep reinforcement learning is used to decide when to shut down radio bands that provide additional capacity to an area. Deep reinforcement learning is also used in [134], to optimise radio and energy resource management in renewable energy-powered wireless networks. A trade-off between energy consumption and Quality of Service (QoS) is investigated in [135], using DL to predict traffic. Hence, in [135] the energy consumption of two different strategies; resources on demand and resources on produced energy, are compared and show how the latter strategy is preferred. The work in [135] compares three techniques for traffic prediction to the baseline; Block Linear Regression (BLR), ANN and Long Short-Term Memory (LSTM).

Energy saving potential: By upgrading the existing 4G or 5G network, several improvements in terms of energy reductions can be seen. The savings estimated by this work translates into 2% of all parts of the RAN excluding the

radio processing which can be reduced by 32%, which is very significant considering its large impact on the overall mobile network energy consumption.

B. ENERGY REDUCTION POTENTIAL BEYOND 5G

By the time of writing this survey, 6G is still in a very early stage, but indeed it will rely on high computational intelligence. The energy improvements enabled by 6G are related to the following topics:

- · Zero touch networks
- Rate Splitting
- Intelligent reflecting surfaces
- Improved sampling techniques
- Energy Harvesting

Zero touch networks are envisioned to ease network operations in the future, by fully automated networks requiring no human interaction. This is achieved by the use of various kinds of AI techniques, since future intelligent network management will depend on the collaboration between the design, deployment and operational phase [114]. Network automation techniques include Self-Organising Network (SON), which uses AI to predict network behavior and control the network based on these findings. Hence, the ultimate goal of the zero touch networking concept is to enable an autonomous network system, which is capable of self-configuration, self-monitoring, self-healing, and self-optimisation based on service-level policies and rules without human intervention [136]. The work in [114] highlights three AI learning methods with great potential for 6G communication:

- Deep reinforcement learning: Using DL models to map from the state to the corresponding action [114].
- Transfer learning: Utilising the constructed knowledge system while solving a problem to the different but related problem [114].
- Federated learning: Decentralising the training process using AI models located on distributed devices [114] without sharing data to a centralised controller. Only model parameters for centralised model training are shared with the centralised controller.

Zero touch networking and AI techniques are used in [44] to optimise the energy consumption of a sliced network, and the same in [137]. The work in [138] proposes a fully automated optimisation and management framework which is claimed to improve energy efficiency by up to 84%, however; it is not stated where in the network the energy efficiency improvements are seen, and it is definitely a very high number. Furthermore, the work in [44] promises a ten-fold increase in energy efficiency gain. However, the work in [139] expresses how it is not possible to gain huge improvements in all metrics using zero touch networking, the example used, is that increasing the number of access points improves the admission rate while incurs more power consumption. The concept of zero touch networks are surveyed very recently in [136].



FIGURE 12. The concept of rate splitting where each users messages are split into a common and a private part. In the RU transmitter, the common parts of multiple messages are combined in one stream.

Zero touch networks are envisioned to improve both energy consumption and CEX.

Rate Splitting is the multiple access technology currently envisioned for 6G communication systems. Rate splitting is an efficient interference mitigation strategy, where each user's message is split into a private part and a common part [140]. Rate splitting splits user messages, into a nonorthogonal transmission of common messages decoded by multiple users and a private message decoded by its corresponding user [141]. The concept of rate splitting is illustrated in Fig. 12. Rate splitting enables partially decoding the interference and partially treating the interference as noise, and thus; perform well at all levels of interference [141]. Hence, the users will decode parts of other users messages and treat the rest as noise. In the transmitter part of the RU, the common messages of multiple users are combined in one stream before the encoder. Then the one combined stream is sent to the user together with the current user's private stream. All users will decode the common part of the message first. Rate splitting results in huge interference improvements leading to continuous improved spectral efficiency resulting in improved energy efficiency [141]. The work in [142] demonstrates an energy efficiency gain of 97% when employing rate splitting instead of simply treating interference as noise combined with the proposed polynomial complexity algorithm. On the other hand, authors in [143] address the practical challenge of the transmitter only knowing the users statistical Channel State Information (CSI). One thing that must be carefully aligned in a rate splitting system is the rate allocation, and the work in [144] shows how careful alignment of the rate allocation can outperform Non-Orthogonal Multiple Access (NOMA) in terms of both energy efficiency and spectral efficiency. Rate splitting is envisioned to have a positive impact on the CEX and a large impact on the energy savings of the mobile network.



FIGURE 13. The concept of Intelligent Reflecting Surfaces (IRS) where IRS is used to redirect the radio signal sent from the base station around an obstacle.

Intelligent Reflecting Surfaces (IRS) are meta surfaces, which consist of many low-cost reflection elements, able to induce real-time changes to the reflected signals [140]. The concept of IRS is illustrated in Fig. 13, where the signal from the antenna is reflected to another direction at the IRS. In practice, IRS consist of a large amount of reflecting diode units, able to reflect any incident electromagnetic signals with an adjustable phase shift [145], [146]. Hence, the signal can be not only reflected in the current angle of reflection, but also in other angles [147]. The reflective elements are considered two-dimensional structures because they have the length of multiple wavelengths, while they are only a fraction of a wavelength deep [147]. Each reflective element is individually adjustable, enabling passive beamsteering, which will result in an increased efficiency of the network [140], [146]. However, in a multi-user scenario, the utilisation of the passive beamsteering will introduce an increase in interference for the users [140]. The work in [140] investigates how rate splitting can mitigate the interference caused by IRS. IRS, will not use any power amplifier to amplify and forward the incoming signal. Thus, an IRS will consume much less energy than a regular relay transceiver [148], furthermore; the gradient descent and fractional programming of IRS will significantly optimise the phase shifts and thus; the transmit power [145]. The work in [149] confirms that IRS will improve the energy efficiency, however; it also shows that increasing the number of reflection elements, will not always ensure the improvement in energy efficiency. Authors in [145] defines IRS as a fundamental enabling technology for 6G with an immense potential toward energy efficiency. The work in [150] compares the energy efficiency of aerial IRS and aerial relaying and notes that the IRS system always outperforms the corresponding relays. Results in [150] show energy efficiency gains up to 74% using IRS. Meta surfaces are envisioned to have a larger positive impact on the CEX and a minor effect on the energy savings.

Sampling will be optimised in 6G mobile networks by introducing compressive sensing techniques. Compressive sensing describes how the signal is reconstructed by taking the samples at a higher rate [148]. More precisely the sampling rate in compressive sensing must be greater than equal to twice the highest frequency of that signal [148]. Compressive sensing is a sampling framework that can be sampled at a rate smaller than the Nyquist rate used today and the resulting, smaller set of samples are sufficient to reconstruct the original signal [148]. However this requires a signal characterised by sparsity and incoherence [148]. Compressive sensing is achieved in a computationally efficient manner, and in compressive sensing, both the sampling and compression is carried out at the same time, which it is not in todays networks [148]. Hence, compressive sensing is also a way to reduce the amount of data that need to be exchanged for required communication [148]. Compressive sensing is already known from sensor networks [151], [152]. Fundamental concepts and widely used algorithms for compressive sensing are comprised in [153]. Improved sampling is envisioned to have a significant positive impact on the CEX and a relative effect on the energy savings.

Energy harvesting is envisioned to extend the range and stand-by times of the mobile network by replacing the conventional ways of powering devices and sensors by tapping the energy from ambient environment [148]. The two broad class of sources for energy harvesting are [148]:

- 1) Natural sources
- 2) Man-made sources

Natural sources of energy harvesting count for example solar and wind power, which is energy supplies in macro scale. On the other hand, man-made sources, which is generating energy supplies in micro scale, such as μW and mW. Hence, energy harvesting from man-made sources is merely used for powering Internet of Things (IoT) devices in all communication networks. However, since a wireless signal can carry and transmit energy and information at the same time, energy harvesting can also prolong the service life of relay devices [154]. Examples of man-made energy harvesting sources count vibration, heat, mechanical movement and RF energy harvesting which is mainly used for wireless communication. RF energy harvesting is a promising technology for powering a huge number of devices, since it eliminates the need for powering those devices from an electric grid and also enables battery lifetime extension [107]. RF energy harvesting describes how to harvest energy from RF waves to extend the network lifetime [148], using the RF range between 300GHz to 3kHz [68]. Nodes that harvest more energy can share their energy with other nodes using energy cooperation [148]. For RF energy harvesting, the Synchronous Wireless Information and Power Transmission (SWIPT) technology is used, which is a combined technique of wireless information transfer and wireless power transfer [155]. Hence SWIPT has two conflicting

goals: maximizing the amount of information transfer versus maximizing the amount of energy transfer [155]. SWIPT has two basic modes: Power Splitting (PS) and Time Switching (TS) [156]. In the receiver, the received signal is divided into an information part and an energy part, thereby the receiver is receiving both information and energy, respectively [156]. In TS, the receiver node separates time to process information and harvests energy, while in PS, the receiver node splits the received signal partly for information processing and partly for the energy harvesting [154]. The practical implementation of RF energy harvesting, requires an optimal balance among the harvested energy and the achievable transmission rates, which can be realised through different approaches [107]. Approaches to finding this optimal balance can be:

- To divide the received signal into two parts, one for energy supply obtained through energy harvesting and the other for information decoding [107]
- To split the information of transmission and energy harvesting in time, to first harvest energy from received signals and then perform wireless information transmission [107].

The realisation of RF energy harvesting requires system performance which is significantly limited by the severe RF signal path loss since this results in low energy conversion efficiency at the position of the energy harvester [107]. Another approach is to harvest energy from the interference. Interference energy harvesting can compensate for the loss of the SWIPT relay system rate caused by the interference [157]. This technique is also based on the existing two operation strategies, TS and PS, where the interference energy is divided among subsequent transmission blocks [157]. Energy harvesting is surveyed in [158], which is referred to for further details. Energy harvesting is not envisioned to have any effect on either the CEX nor the energy consumption, however; it will play an important part in the journey to CO2 neutrality.

Energy saving potential: The energy saving methods predicted for 6G are examined throughout this subsection. The solutions and proposals elaborated in this section are not yet available and thus; they will not be considered in the final analysis in Section VII. However, the techniques proposed for future mobile network generations are foreseen to provide significant improvements on the energy efficiency.

C. THE ROAD TO CO2 NEUTRALITY

So far, this work has examined many different proposals for minimising the energy consumption of the mobile network. However, the mobile network will always consume energy and thus; methods toward achieving CO2 neutrality are briefly considered in this subsection. These methods count:

- Alternative power sources
- Heat re-use
- Modern Batteries
- Liquid cooling

Alternative power sources are most widely known as solar panels, water- and wind power. Thus, if the mobile network is utilising alternative power sources when available, it will become CO2 neutral in that period of time. When the alternative energy sources are producing more energy than needed to run the network, energy can be saved on batteries for later use. The work in [159] utilises solar panels to power base stations and energy storage as well, to obtain carbon footprint reductions at the order of 50%. The work in [135] investigates power saving opportunities using ML predictions for a base station both equipped with a photovoltaic panel and connected to the power grid. Results show energy savings between 10% and 40% [135]. Most of the energy is produced locally by the photovoltaic panels, and only a small amount, between 8% and 23% of the energy, is purchased from the power grid, this is typically during night, when energy is cheaper [135]. The paper [135] does not state which components of the network achieve these large power savings, however; it is interesting that only up to 23% of the energy is provided by the power grid. However it must be noted that measurements were carried out in the sunny Milano, Italy [135].

Heat re-use can efficiently be used to improve the sustainability of the network. Datacenters generate heat, no matter if it is - in mobile network context, a larger datacenter carrying out baseband processing from a large number of C-RAN sites, or a small on-site datacenter only handling the baseband processing of the specific site. Hence, re-using the heat for warming up buildings will be an efficient way to save energy in other sectors. The work in [160] investigates how datacenters' waste heat can be re-used to warm up buildings, which is in [161] considered of much higher economic value than the thermodynamic value of the heat.

Modern Batteries can be of various types, for example flow batteries and lithium batteries. Lithium batteries are becoming an integral part of 5G sites to enhance energy management [23]. Flow batteries use cells in liquid form where the energy-storing materials flow in-between while undergoing redox reactions [162]. Hence, flow batteries are different from solid state batteries in terms of flow structure and decoupled power and energy [162]. Modern batteries are used to harvest energy at the right times, when alternative power sources are available, for example when the weather is sunny or windy, and store the energy until the day where its cloudy and no wind.

Liquid cooling is being implemented to reduce the need for air conditioning [23]. The concept uses cold liquids to cool down the heat generating equipment. The liquid is heated up by the equipment and will be led away to be cooled down by air, before the cyclic system leads it back again to cool the equipment once again.

VI. SHARING THE NETWORK

Mobile networks consume a large amount of energy. Hence, in principle it is a waste of resources when more than one network cover the same area. This is the reason why the



FIGURE 14. An illustration of two individual networks compared to using different types of network sharing: MORAN, MOCN and network slicing.

concept of network sharing is relevant seen from an energy consumption minimisation perspective. However, multiple mobile networks covering the same area do also provide significant benefits in terms of reliability, avoiding monopoly and thus; obtain better prices for the customers. A shared network is a totally different business case than most operators use today, but it raises the question: Why have multiple networks covering the same piece of land, when in other sectors (sewer, power grid, traffic, i.e.,) only one network is necessary? This section will focus on different techniques for multiple operators to share the same network. For RAN sharing the two most commonly used solutions are Multi-Operator Core Networks (MOCN) and Multi-Operator RAN (MORAN). However, another approach is to use the concept of network slicing to create a multi tenant network. On the other hand, it is also possible for an operator rent capacity on another operators network, defined as roaming. The four approaches are defined below and illustrated in Fig. 14.

- Roaming: A network operator rents capacity on another network operator's network.
- MORAN: The whole RAN (antenna, cell site, power), except the radio frequencies, is shared between multiple operators.

- MOCN: Multiple core networks share the same RAN including the radio frequencies.
- Network slicing: Multiple tenants share or rent a slice of one network where resources are assigned based on an individual basis, which could be the amount of owned radio spectrum.

A. ROAMING

In a roaming scenario, traffic is moved from one network to another. Hence, from an energy saving perspective it might be cheaper, under specific circumstances, to buy resources in another network rather than paying for the energy consumption in ones own network. Furthermore, there might be potential energy savings in low traffic periods, if only one network is running in the time period and the remaining networks roam in case of any traffic. The concept is investigated in [163] where operators achieve up to 11% energy savings by deactivating their own base stations and roaming on other networks. The work in [164] evaluates different scenarios where a number of network operators can share one network in low utilisation periods. In general roaming is foreseen to bring a reduction in the energy consumption, since users are handled by another network which frees up time for enhanced sleep modes or deactivation of whole sites. Roaming can even bring improved resource utilisation, in case users in an area are gathered to be roaming on the same network. This could for example be in low traffic periods. The CEX impact of roaming will depend on acquired Service Level Agreement (SLA).

B. MULTI OPERATOR NETWORK

If the networks from multiple operators shall be merged, it must be taken into account that the different operators under some circumstances have different cell site positions. Hence, the total amount of site positions will increase in order to keep the same level of coverage and capacity to all operators' customers. Figure 14 illustrates how two operators can share the same network, either as MORAN or MOCN. The major difference between these two concepts is that in MOCN, everything is shared equally between the two operators despite from them having individual core networks. However, in MORAN the two operators operate on their own radio frequencies. Thus, if one own more spectrum than the other, then the operator with more spectrum will also have more capacity.

MOCN enables equal sharing of the RAN between multiple operators. MOCN is standardised by 3GPP in [165] and [166]. When two operators share the same RAN, scheduling fairness among the users served by the different operators is an important matter [167]. In [167] the scheduling fairness is evaluated based on different parameters like number of scheduling Physical Resource Block (PRB)s per slot, number of scheduling times per second and throughput. The RAN sharing of different operators' users can be realised by simultaneously broadcasting the Public Land Mobile Network (PLMN) IDs of all operators in two shared frequency bands, then users can access any of the shared frequency bands [167]. The work in [168] defines and verifies three key features of MOCN: Multiple PLMNs, inter-frequency handovers and prioritisation of different frequencies. Compared to MORAN, then the MOCN solution is more efficient also in terms of cost as well as it enables a better utilisation of the shared spectrum [169]. In Denmark, the TT Network uses MOCN configuration [170], but MOCN is also used in for example Sweden and Australia [171].

MORAN is the mainly adopted solution for network sharing due to its guaranty of capacity independence and service differentiation among the RAN sharing operators [169]. Hence, operators can independently control cell level operations like deciding their own optimisation parameters and transmission power to control the cell range as well as manage interference [172]. However, not much work relates to the use of MORAN alone, it is merely related to the use of network slicing which will be described under Multi Tenant Network. In Japan, the operators SoftBank Corp. and KDDI are deploying a shared network using MORAN technology [173], and it is already used in for example Spain or United Kingdom (U.K.) [171].

Multi-operator networking energy savings: Energy savings in a multi-operator network will depend on the amount of operators sharing the network. In this section, as an example, three networks are merged into one. Power consumption from RUs and air conditioning are expected to be reduced by 66% on a per site level, however; it is assumed that the operators will not share the exact same site positions. Thus, the number of sites will grow, but overall a large energy saving is expected in the range of 50%. The main control and baseband processing will save energy consumption based on the amount of improved resource utilisation, since the amount of traffic and number of users are the same. It is expected that 30% is a realistic saving due to improved utilisation of traffic fluctuations. A reduction in the power supply energy consumption is also expected since less components are supported, in the area of 10% since existing equipment will use more energy. From a CEX perspective, assuming the network is dimensioned after the new amount of users, MORAN is not expected to bring any CEX changes. However, MOCN users will be positively or negatively impacted depending on their operator's previous resource capacity and configurations.

C. MULTI TENANT NETWORK

Network slicing describes the ability of running multiple separate logical networks on top of one shared physical network infrastructure. Hence, one physical network composed of antenna masts, processing units and transport network connections will be able to handle individual logical networks with individual performance requirements. These logical networks are defined as "slices" and one or more slice can be rented by a tenant. Network slicing enables the opportunity of isolating different services optimised for a specific purpose, improving the performance compared to the traditional "one-size-fits-all" network. From an energy consumption perspective, the use of network slicing will enable only one network operator handling the network of multiple tenants and, using the country of Denmark as an example, reduce the number of networks from three main operators and a number of IoT and private network operators, to one physical network. Hence, network slicing requires from all nodes in the network to be able to separate individual data streams and provide differentiated QoS. The end-to-end network infrastructure for network slicing is issued in [174], [175], [176], [177]. The network slicing control framework layer is comprised of radio slice controllers, transport slice controllers and core slice controllers, controlling the RAN, crosshaul and core domains respectively [175]. In [178], the energy consumption of network slicing is investigated for two approaches; slice-based resource allocation and user-based resource allocation. The work in [178] illustrates how endto-end network slicing provides larger decreases in energy consumption compared to slicing only the RAN. In this case up to 20% energy savings are achieved and up to 30% less bandwidth in a low load scenario for end-to-end slicing [178]. Hence, the implementation of network slicing requires all parts of the mobile network to be able to isolate traffic flows, including:

- Radio functions
- Baseband processing
- Crosshaul Transport
- Core network

The following part will analyse different options for network slicing implementation.

Radio Functions provide capacity based on the amount of spectrum owned by the individual operator. Following 3GPP [80], each slice may be assigned to either shared or dedicated radio resources. Hence, different methods are proposed:

- Each slice has a fixed, predefined contiguous sub-band available [179]
- Each slice has a contiguous sub-band consisting of a fixed region and a variable region available [179]
- The resource grid is divided into regular sub-bands of fixed size, which can be assigned to different slices on demand [179]
- Assigning different Transmission Time Intervals (TTI)s to different slices, hence; a fixed time where each slice can transmit. This way, a certain number of PRBs will be assigned to each slice, enabling guaranteed performance [180]
- Dynamical reallocation of PRBs [169], [181]
- Scheduling a number of slots for each slice [175], [182], [183]

Baseband processing includes the very important MAC scheduler. The MAC scheduler can assign capacity dynamically to different slices and simultaneously keep track of the individual guaranties assigned to the different slices. The scheduler will also be responsible for traffic prioritisation

in the core- and crosshaul networks. Three implementation options for a network slicing compatible MAC scheduler can be derived:

- One joint scheduler with specific rules for each network slice. This requires a MAC layer of high complexity to manage multiple different slices configurations demands. However, this one scheduler will have an overview of the entire system, enabling more efficient sharing of unused resources [179]
- Individual schedulers for individual slices. The individual schedulers require sufficient coordination inbetween. However, the schedulers are simpler and provide a more flexible customisation of each individual network slice [179]
- One master MAC which allocates resources between slices and multiple customised slave MACs to schedule the resources within the different slices [175]

Crosshaul Transport is the infrastructure connecting the RAN to the core. The transport network infrastructure, used in a network slicing enabled network, has to carry the traffic from several logical networks, each having very different requirements to the one physical network. Hence, the crosshaul network must support separation of different flows and slice isolation in all crosshaul network nodes. Thus, options to be used for network slicing, for different types of crosshaul nodes implementation includes:

- Ethernet: Allows sharing of the network infrastructure through standardised virtualisation techniques [184], referred to as Virtual Local Area Network (VLAN) [185]. Thus, Ethernet can flexibly allocate and dynamically reserve resources for specific slices [81]. Using Ethernet, the physical resources can be highly utilised, as Ethernet flexibly allocates capacity within the fibers and the switches [81].
- Wave Division Multiplexing (WDM): Transports data from different sources on different wavelengths. Using WDM, it is possible for each slice to operate on its own wavelength. Wavelengths are isolated, and thus; they can carry individual QoS performance requirements. However, all traffic have to go to the same endpoint, and it is not possible to assign extra capacity on demand.
- Passive Optical Network (PON): Aggregate data to the same link in the Optical Line Terminal (OLT) and requires uplink and downlink traffic to be handled differently. In [186], the use of PON as a transport medium for network slicing is demonstrated using VLAN and a new control plane to improve the flexibility of resource management and enable creation of different slices on demand. The uplink transmission is demonstrated in [187] proposing isolation of traffic from different slices inside the OLT. This enables the network to handle different amounts of data from different sources and network slices will have access to the uplink transmission is demonstrated using a scheduler, to ensure required and isolated QoS for the different slices [189].

The Core network is where all user requests are processed. The 5G core network is a service-based architecture, in which network functions cooperate to perform signaling procedures, such as connection, registration, mobility management, and session management [190]. Slicing the core network involves specifying network services according to the functional and quality requirements [190]. Hence, each slice is composed of a set of core network resources and is assigned to one or more services [191]. When multiple network slices are supported by the core network, a network slice selection function helps in effective slice instance selection during the UE registration and session establishment to an access management function [192]. The core network has VNFs classified as common network functions and special network functions [175]. The common network functions are shared by different slices, and the special network functions are dedicated for the related slice [175].

Multi-tenant networking energy savings: Compared to the savings achieved in the multi-operator network, then the multi-tenant network will additionally save energy in the RAN infrastructure and the core network. Infrastructure savings are estimated to result in 30% since much more traffic has to be carried by one network, and thus; expansions are expected to be necessary. From a CEX perspective, multi-tenant networking is not expected to bring any negative impact, it might be a positive impact due to the specialised handling of different use cases.

VII. DISCUSSION AND GUIDELINES

This work provided multiple ways for a network operator to reduce their energy consumption, but the big question is; where to begin? This section will elaborate on the proposals provided in the previous sections and assess which savings will matter the most, based on which part of the network they contribute to. Hence, first a per-unit overview is provided and then a set of guidelines directing more implementation specific goals.

A. PER-UNIT ENERGY REDUCTIONS

This subsection provides an overview of all parts of the RAN presented in Section III and how to improve the energy efficiency in the specific part, based on the lessons learned from Sections IV to VI.

Radio Processing is represented by the RU, which depending on the functional split used, will contain more or less functions. However, what is always present in the RU, is the power amplifier, which is the most energy consuming part of the RU [4]. Thus, from a radio processing perspective, the changes with the highest impact on the energy consumption will concern the amplifier. The amplifier's energy consumption can be minimised by reducing the distance between RU and antenna, use multi-band RUs, GaN materials and sleep modes such as symbol shut down or mMIMO muting. Other ways to improve the energy consumption of the RU can be to choose the most energy efficient RU on the market - enabled by open RAN; to share one network between different operators, minimising the total amount of RUs; or

Air conditioning may vary in its impact of energy consumption, by the capacity of the air conditioning units and the climate [4] but also by the seasonal temperatures. However, in [4] air conditioning is stated to contribute to a large amount of the base station energy consumption. The air conditioning energy consumption can be reduced by gathering all baseband processing servers in datacenters, following the C-RAN architecture. Another way to reduce air condition energy consumption is to re-use the heat generated by the servers for other purposes. Thus, the sum of estimated savings proposed in Sections IV and V is around 52%, but this number is highly depending on whether and how much air conditioning is used.

Baseband processing represents a minor share of the total base station energy consumption. Methods to improve the baseband processing energy consumption is to use the most energy efficient HW; workload scheduling and load consolidation enabled by vRAN and C-RAN; or to introduce HW sleep states. The sum of estimated savings proposed for the baseband processing in Sections IV and V is around 42%.

Main control energy consumption improvement techniques include to choose the most energy efficient HW; use HW sleep states; or reduced inter-BBU communication overhead. The estimated savings proposed for the main control in Sections IV and V is around 22% utilising coordinated transmission, improved resource utilisation and HW sleep states.

Power supply energy consumption, can be reduced if multiple operators share the same network or if a carrier is shut down. The sum of estimated savings proposed in Sections IV and V is around 12%.

Estimated energy reduction potential: The outcome of energy saving potentials listed in this section are illustrated in Fig. 15. The figure shows how the energy consumption of the mobile network can be reduced by up to 35% using methods suggested in this work. The baseline to achieve these large savings is a network using a proprietary installation of distributed RAN, with no sleep modes, processing resource utilisation and less efficient power amplifiers, ie. none of all what was described earlier in this section. The contributions from Sections IV-VI to this overall saving are listed in Table 2. Hence, Table 2 lists the energy consumption savings for the different components, which using the model in Section III provides a saving of more than 30% of the overall mobile network energy consumption. This number is only including architectural and technological contributions, which are listed in column "total". Furthermore, great potential for energy savings is seen from Section VI, where a large amount of the RAN energy consumption can be saved by



FIGURE 15. The energy consumption of the mobile network in the inner circle, showing 48% energy consumption savings in the RAN, corresponding to 35% of the total mobile network energy consumption. The middle circle shows the share of the RAN, divided out on base stations and infrastructure. The outer circle shows the base station components.

TABLE 2. Sum of estimated potential energy savings per base station component.

RAN component	Architecture	Technology	Total	Share of Mobile Network Energy Consumption Saving
Main control	20%	2%	22%	1%
Baseband processing	40%	2%	42%	2%
Power supply	10%	2%	12%	0.5%
Air condition	50%	2%	52%	15%
Radio processing	25%	32%	57%	16.5%

sharing the network with other operators. The exact number will, however; depend on a granularity of factors including the number of operators sharing the physical network.

B. GUIDELINES TOWARDS GREEN RAN

How will the future, energy optimised mobile network look like? This subsection provides guidelines for initial smaller changes with a large impact and then discusses the direction to move in for a brownfield operator and design implications for a greenfield operator.

1) INITIALLY

In a smaller and near-future timescale it is beneficial to look into changes that will show the largest possible impact with the least amount of work, the low-hanging fruits so to speak. Hence, a starting point is to consider optimisations of the RU, that can be incorporated as software updates. These optimisations count:

- Sleep modes
- AI predictions to optimise sleep modes
- · Periodical roaming



FIGURE 16. To the left is an illustration of the current network deployment for many operators. To the right is a proposal of the future energy optimised network architecture, exploiting the concepts of: C-RAN, vRAN and network sharing using network slicing. Hence, the baseband processing is carried out in software running on Commercial Off The Shelf (COTS) Hardware (HW) which is centralised in datacenters.

Following above recommendations will bring large energy savings to the network, but using minor changes. When equipment is changed any way, or new equipment is deployed, it will be beneficial to look into the following technologies:

- Multi-band RUs
- · GaN amplifiers

This way, the network follows the energy optimised track and is prepared for future architectural changes. Another thing to introduce in the network, which will probably not bring the big energy savings right away, but rather prepare the network for future demands is:

• vRAN

2) LATER

In order to optimise the future mobile network in terms of energy usage, a completely new architectural design will result in the most significant impact. Hence, a proposal is illustrated in Fig. 16, utilising the concepts of:

- C-RAN
- · Network sharing
- Open RAN

Paving the way towards the mobile network of the future, many operations can be executed by the operator self, and introducing C-RAN in a network already enabled by vRAN, the network operator must be mindful of; fronthaul network and cloud control. Furthermore, it is important to evolve a deployment strategy, determining if the CU and DU functions should be handled in the same datacenter or at different distances.

Moving to a shared network or to open RAN requires collaboration with more stakeholders. Moving to a shared network solution requires a number of network operators to agree on collaborating and corresponding terms and conditions. Moving to an open RAN solution requires for the network operator to find the right products and partnerships as well as finding a way to manage and orchestrate the multi-vendor network.

Figure 16 shows how the current network of today is designed in many countries; each network operator run their own network and BB processing is not shared. The proposal of the future energy optimised network architecture design, to the right in Fig. 16, utilises visions of shared RAN and shared BB processing. Furthermore, the SW is decoupled from the HW opening up the opportunity of choosing the most energy efficient equipment available and a much higher degree of network control. A centralised network control, which can be supported by AI techniques will optimise functions such as load control, load consolidation, dynamic processing selection and sleep modes. Furthermore, opportunities for achieving CO2 neutrality, as presented in Section V, can make a great change and lead the network operator towards a green, CO2 neutral future.

Changes toward an energy consumption minimised future RAN presented in this work are many-fold and in some cases they also point in opposite directions. Thus, choices to make as a network operator are not always straight forward, and changes can appear at many different levels. Figure 17 illustrates, the pains and gains of a network operator in terms of CEX and energy savings. Furthermore, the technologies presented are represented by how much effort their implementation require, this is illustrated by different colours. The graph shows in the upper right part technologies that will both improve CEX and energy consumption. In the lower right part of the graph are technologies that will not improve CEX, but will improve the energy consumption. Hence, introducing "sleep modes", will lead to degraded CEX, as they add delay or reduce capacity, however; they improve the energy savings and they are fairly easy to implement. Thus, sleep modes are placed in a negative position on the CEX axis, in a positive position on the energy saving axis and coloured green for minimal effort. On the other hand introducing "zero touch networks", requires a medium effort, because it is a complex process, maybe even tangible to large installation effort. However, zero touch networks will to a large extend optimise the networks and thus; is placed on a very positive position of both the CEX axis and the energy savings axis.

3) GREENFIELD GOING GREEN

Being a greenfield operator, starting from "the scratch", means that there is no legacy to be mindful about. Hence, a greenfield operator can choose the most modern and energy efficient way to deploy the network. Thus, greenfield installations conducted within the latest years use both C-RAN, vRAN and open RAN architectures. As illustrated in the right side of Fig. 16. Examples of network operators deploying



FIGURE 17. Pain and gain map over Customer Experience (CEX) and energy savings. Hence, the upper right part of the figure shows technologies that benefits both energy and CEX. Correspondingly, the lower right part of the figure shows the technologies that will benefit the energy consumption but at the cost of lower CEX. Technologies are marked by different colours depending on the implementation effort they require.

greenfield networks within the latest years count DISH in the U.S. [193], Rakuten in Japan [194], [195] and 1&1 in Germany [196]. However, these greenfield operators are rolling out their installations in countries where multiple other networks already exist, therefore; one may think, is yet another network actually necessary? On the other hand, these new and modern installations can potentially allow for future tenants, to minimise the footprint of mobile networks.

VIII. CONCLUSION

This work outlined different technological and architectural improvements that will minimise the energy consumption of future mobile networks. Furthermore, current research directions and visions for green 6G were examined. Different proposals for energy reductions were analysed and their contributions were compared and evaluated in the perspective of their energy consumption minimisation potential for future mobile networks. It was shown how future network architectures can be energy optimised by exploring the potential of new architectures, network sharing opportunities and features enabled by emerging technologies in different areas. Hence, by highlighting research trends in areas like power amplifier minimisation and resource utilisation. Opportunities identified by this work are proposed together with concrete deployment methods. Results demonstrate an overview of the different energy minimisation methods examined in relation to their overall energy reduction contributions. Results show that implementing selected features and architectures, the mobile network overall energy consumption can be reduced by approximately 30%. Following this, a set of guidelines for the brownfield network operator's road towards an energy optimised mobile network was provided. These guidelines proposed initial changes and changes for the longer run as well as a roadmap over pains and gains related to CEX and energy savings. Finally a vision for the future energy optimised mobile network architecture was presented utilising the concepts of C-RAN and multi-tenant networks.

ACKNOWLEDGMENT

A special thanks to colleagues from TDC Net: Henrik Hansen, Thomas Vilhelmsen, Szymon Izydorek and Peter Beltoft, as well as from DTU Electro: Mingyuan Zang and Kartikeya Sharma, for sparring and feedback during the process of writing this survey.

ACRONYMS

3GPP	3rd Generation Partnership Project.
ACK	Acknowledgement.
ADC	Analogue-to-Digital Converter.
AI	Artificial Intelligence.
ANN	Artificial Neural Networks.
ASM	Advanced Sleep Mode.
BB	Baseband.
BBU	Baseband Unit.
BLR	Block Linear Regression.
C-RAN	Cloud RAN.
CEX	Customer Experience.
CNF	Cloud-Native Network Function.
CoMP	Coordinated Multipoint.
COTS	Commercial off the Shelf.
CPRI	Common Public Radio Interface.
CPU	Central Processing Unit.
CSI	Channel State Information.
CU	Centralised Unit.
DL	Deep Learning.
DU	Distributed Unit.
ETSI	European Telecommunications Standards
2101	Institute
GaN	Gallium Nitride
GSMA	GSM Association
HARO	Hybrid Automatic Repeat Request
нς	High Lover Split
HW	Hardware
	Information and Communications Technology
	Internet of Things
101	In phase Quadrature
IQ IDC	Intelligent Deflecting Surfaces
IKS	Ken Derformen er Indiaster
	Levy France Schedular Solution
LESS	Low Energy Scheduler Solution.
LLS	Low Layer Spill.
LSIM	Long Short-Term Memory.
MANO	Management and Orchestration.
MBH	Mobile Backhaul.
MDAF	Management Data Analytics Function.
MIMO	Multiple Input Multiple Output.
ML	Machine Learning.
mMIMO	massive MIMO.
mmwaves	milimeter waves.
MOCN	Multi-Operator Core Networks.
MORAN	Multi-Operator RAN.
NFV	Network Function Virtualisation.
NGMN	Next Generation Mobile Networks.
NOMA	Non-Orthogonal Multiple Access.
NWDAF	Network Data Analytics Function.
OFDM	Orthogonal Frequency Division Multiplexing.
OLT	Optical Line Terminal.
PLMN	Public Land Mobile Network.
PON	Passive Optical Network.
PRB	Physical Resource Block.
PS	Power Splitting.
QoS	Quality of Service.

RAN	Radio Access Network.
RATs	Radio Access Technologies.
RF	Radio Frequency.
RIC	Radio Intelligent Controller.
RU	Radio Unit.
SCMA	Sparse Code Multiple Access.
SFP	Small Formfactor Pluggable.
SLA	Service Level Agreement.
SNR	Signal to Noise Ratio.
SON	Self-Organising Network.
SW	Software.
SWIPT	Synchronous Wireless Information and Power
	Transmission.
TIP	Telecom Infra Project.
TS	Time Switching.
TTI	Transmission Time Intervals.
UE	User Entity.
UK	United Kingdom.
VLAN	Virtual Local Area Network.
VNF	Virtual Network Function.
vRAN	virtual RAN.
WDM	Wave Division Multiplexing.

REFERENCES

- [1] C. Freitag, M. Berners-Lee, K. Widdicks, B. Knowles, G. S. Blair, and A. Friday, "The real climate and transformative impact of ICT: A critique of estimates, trends, and regulations," *Patterns*, vol. 2, no. 9, Sep. 2021, Art. no. 100340.
- [2] H. Ritchie, M. Roser, and P. Rosado. "CO₂ and Greenhouse Gas Emissions." 2020. [Online]. Available: https://ourworldindata.org/ co2-and-other-greenhouse-gas-emissions
- [3] Ericsson Mobility Report June 2022, Ericsson, Stockholm, Sweden, 2022.
- [4] Green Future Networks: Network Energy Efficiency, NGMN Alliance, Frankfurt, Germany, 2021.
- [5] M. Masoudi et al., "Green mobile networks for 5G and beyond," *IEEE Access*, vol. 7, pp. 107270–107299, 2019.
- [6] N. I. Sarkar, D. P. Singh, and M. Ahmed, "A survey on energy harvesting wireless networks: Channel capacity, scheduling, and transmission power optimization," *Electronics*, vol. 10, no. 19, p. 2342, 2021. [Online]. Available: https://www.mdpi.com/2079-9292/10/19/2342
- [7] D. A. Temesgene, J. Núñez-Martínez, and P. Dini, "Softwarization and optimization for sustainable future mobile networks: A survey," *IEEE Access*, vol. 5, pp. 25421–25436, 2017.
- [8] F. Marzouk, J. P. Barraca, and A. Radwan, "On energy efficient resource allocation in shared RANs: Survey and qualitative analysis," *IEEE Commun. Surveys Tuts.*, vol. 22, no. 3, pp. 1515–1538, 3rd Quart., 2020.
- [9] A. Israr, Q. Yang, and A. Israr, "Power consumption analysis of access network in 5G mobile communication infrastructures—An analytical quantification model," *Pervasive Mobile Comput.*, vol. 80, Feb. 2022, Art. no. 101544.
- [10] M. F. Hossain, A. U. Mahin, T. Debnath, F. B. Mosharrof, and K. Z. Islam, "Recent research in cloud radio access network (C-RAN) for 5G cellular systems—A survey," *J. Netw. Comput. Appl.*, vol. 139, pp. 31–48, Aug. 2019.
- [11] A. Checko et al., "Cloud RAN for mobile networks—A technology overview," *IEEE Commun. Surveys Tuts.*, vol. 17, no. 1, pp. 405–426, 1st Quart., 2015.
- [12] N. S. Saad, W. N. S. F. W. Ariffin, J. Nordin, A. S. M. Zain, and L. I. M. Asri, "A literature survey on cloud-RAN architecture toward 5G green wireless communication systems," in *Proc. 1st Int. Conf. Eng. Technol. (ICoEngTech)*, 2021. [Online]. Available: https://doi. org/10.1088/1742-6596/1962/1/012036

- [13] M. A. Habibi, M. Nasimi, B. Han, and H. D. Schotten, "A comprehensive survey of RAN architectures toward 5G mobile communication system," *IEEE Access*, vol. 7, pp. 70371–70421, 2019.
- [14] V. S. Pana, O. P. Babalola, and V. Balyan, "5G radio access networks: A survey," Array, vol. 14, Jul. 2022, Art. no. 100170.
- [15] L. M. P. Larsen, A. Checko, and H. L. Christiansen, "A survey of the functional splits proposed for 5G mobile crosshaul networks," *IEEE Commun. Surveys Tuts.*, vol. 21, no. 1, pp. 146–172, 1st Quart., 2018.
- [16] M. Hoque et al., "Green communication in 6G," in Proc. 8th Int. Conf. Mechatron.Eng. (ICOM), 2022, pp. 101–106.
- [17] M. Alsabah et al., "6G wireless communications networks: A comprehensive survey," *IEEE Access*, vol. 9, pp. 148191–148243, 2021.
- [18] Q. Long, Y. Chen, H. Zhang, and X. Lei, "Software defined 5G and 6G networks: A survey," *Mobile Netw. Appl.*, vol. 27, no. 5, pp. 1792–1812, 2022.
- [19] M. N. Mahdi, A. R. Ahmad, Q. S. Qassim, H. Natiq, M. A. Subhi, and M. Mahmoud, "From 5G to 6G technology: Meets energy, Internetof-Things and machine learning: A survey," *Appl. Sci.*, vol. 11, no. 17, p. 8117, 2021.
- [20] A. Arnaz, J. Lipman, M. Abolhasan, and M. Hiltunen, "Toward integrating intelligence and programmability in open radio access networks: A comprehensive survey," *IEEE Access*, vol. 10, pp. 67747–67770, 2022.
- [21] W. Azariah, F. A. Bimo, C.-W. Lin, R.-G. Cheng, R. Jana, and N. Nikaein, "A survey on open radio access networks: Challenges, research directions, and open source approaches," 2022, arXiv:2208.09125.
- [22] T. Pamuklu, M. Erol-Kantarci, and C. Ersoy, "Reinforcement learning based dynamic function splitting in disaggregated green open RANs," in *Proc. IEEE Int. Conf. Commun.*, 2021, p. 6.
- [23] D. Lopez-Perez et al., "A survey on 5G radio access network energy efficiency: Massive MIMO, lean carrier design, sleep modes, and machine learning," *IEEE Commun. Surveys Tuts.*, vol. 24, no. 1, pp. 653–697, 1st Quart., 2022.
- [24] F. Salahdine, J. Opadere, Q. Liu, T. Han, N. Zhang, and S. Wu, "A survey on sleep mode techniques for ultra-dense networks in 5G and beyond," *Comput. Netw.*, vol. 201, Dec. 2021, Art. no. 108567.
- [25] L. Song, X. Hu, G. Zhang, P. Spachos, K. N. Plataniotis, and H. Wu, "Networking systems of AI: On the convergence of computing and communications," *IEEE Internet Things J.*, vol. 9, no. 20, pp. 20352–20381, Oct. 2022.
- [26] I. P. Chochliouros et al., "Energy efficiency concerns and trends in future 5G network infrastructures," *Energies*, vol. 14, no. 17, p. 5392, 2021.
- [27] T. Huang, W. Yang, J. Wu, J. Ma, X. Zhang, and D. Zhang, "A survey on green 6G network: Architecture and technologies," *IEEE Access*, vol. 7, pp. 175758–175768, 2019.
- [28] D. Feng, C. Jiang, G. Lim, L. J. Cimini, G. Feng, and G. Y. Li, "A survey of energy-efficient wireless communications," *IEEE Commun. Surveys Tuts.*, vol. 15, no. 1, pp. 167–178, 1st Quart., 2013.
- [29] "Service-Oriented Optimization of Green Mobile Networks." 2015. Accessed: Aug. 8, 2022. [Online]. Available: https://soogreen. eurestools.eu/
- [30] W. Labidi, T. Chahed, and S.-E. Elayoubi, "Optimal energy management strategies in mobile networks powered by a smart grid," in *Proc. IEEE Int. Conf. Commun. (ICC)*, 2017, pp. 1–6.
- [31] A. Alabbasi, X. Wang, and C. Cavdar, "Optimal processing allocation to minimize energy and bandwidth consumption in hybrid CRAN," *IEEE Trans. Green Commun. Netw.*, vol. 2, no. 2, pp. 545–555, Jun. 2018.
- [32] D. Renga, H. Al Haj Hassan, M. Meo, and L. Nuaymi, "Energy management and base station on/off switching in green mobile networks for offering ancillary services," *IEEE Trans. Green Commun. Netw.*, vol. 2, no. 3, pp. 868–880, Sep. 2018.
- [33] "Converged Heterogeneous Advanced 5G Cloud-RAN Architecture for Intelligent and Secure Media Access." 2015. Accessed: Aug. 8, 2022. [Online]. Available: https://www.charisma5g.eu/
- [34] UNIVERSITY OF KENT. "Intelligent Converged Network Consol-Idating Radio and Optical Access Around User Equipment." 2015. Accessed: Aug. 8, 2022. [Online]. Available: https://cordis.europa. eu/project/id/644526

- [35] K. Wang and K. Yang, "Resource allocation between service computing and communication computing for mobile operator," in *Proc. IEEE Int. Conf. Commun. (ICC)*, 2017, pp. 1–6.
- [36] H. Mei, K. Wang, and K. Yang, "Multi-layer cloud-RAN with cooperative resource allocations for low-latency computing and communication services," *IEEE Access*, vol. 5, pp. 19023–19032, 2017.
- [37] K. Wang, K. Yang, and C. S. Magurawalage, "Joint energy minimization and resource allocation in C-RAN with mobile cloud," *IEEE Trans. Cloud Comput.*, vol. 6, no. 3, pp. 760–770, Jul.–Sep. 2018.
- [38] C. Pan, H. Zhu, N. J. Gomes, and J. Wang, "Joint user selection and energy minimization for ultra-dense multi-channel C-RAN with incomplete CSI," *IEEE J. Sel. Areas Commun.*, vol. 35, no. 8, pp. 1809–1824, Aug. 2017.
- [39] C. Pan, H. Zhu, N. J. Gomes, and J. Wang, "Joint precoding and RRH selection for user-centric green MIMO C-RAN," *IEEE Trans. Wireless Commun.*, vol. 16, no. 5, pp. 2891–2906, May 2017.
- [40] S. Mao, S. Leng, K. Yang, X. Huang, and Q. Zhao, "Fair energyefficient scheduling in wireless powered full-duplex mobile-edge computing systems," in *Proc. IEEE GLOBECOM Global Commun. Conf.*, 2017, pp. 1–6.
- [41] Centre Tecnologic de Telecomunicacions de Catalunya. "5G Technologies for Reconfigurable and Efficient Radio Access Infrastructure and Mobile Devices." 2019. Accessed: Aug. 8, 2022. [Online]. Available: https://www.cttc.cat/project/5g-technologies-forreconfigurable-and-efficient-radio-access-infrastructure-and-mobiledevices/
- [42] "E2E-Aware Optimizations and Advancements for the Network Edge of 5G New Radio." 2017. Accessed: Aug. 8, 2022. [Online]. Available: https://one5g.eu/
- [43] MonB5G. "MonB5G Project." 2019. Accessed: Nov. 1, 2022. [Online]. Available: https://www.monb5g.eu/
- [44] H. Chergui et al., "Zero-touch AI-driven distributed management for energy-efficient 6G massive network slicing," *IEEE Netw.*, vol. 35, no. 6, pp. 43–49, Nov./Dec. 2021.
- [45] L. A. Garrido, P. V. Mekikis, A. Dalgkitsis, and C. Verikoukis, "Context-aware traffic prediction: Loss function formulation for predicting traffic in 5G networks," in *Proc. IEEE Int. Conf. Commun.*, 2021, p. 6.
- [46] "Network Intelligence for Adaptive and Self-Learning Mobile Networks." 2021. Accessed: Aug. 8, 2022. [Online]. Available: https://h2020daemon.eu/
- [47] J. A. Ayala-Romero, A. Garcia-Saavedra, X. Costa-Perez, and G. Iosifidis. "EdgeBOL: Automating Energy-Savings for Mobile Edge AI." 2021. [Online]. Available: https://doi.org/10.1145/ 3485983.3494849
- [48] J. A. Ayala-Romero, A. Garcia-Saavedra, X. Costa-Perez, and G. Iosifidis, "Orchestrating energy-efficient vRANs: Bayesian learning and experimental results," *IEEE Trans. Mobile Comput.*, early access, Oct. 29, 2021, doi: 10.1109/TMC.2021.3123794.
- [49] J. Ayala-Romero, A. Garcia-Saavedra, X. Costa-Perez, and G. Iosifidis. "Demonstrating a Bayesian Online Learning for Energy-Aware Resource Orchestration in vRANs." Jun. 2021. [Online]. Available: https://doi.org/10.5281/zenodo.4966737
- [50] "Dynamic Coverage Extension and Distributed Intelligence for Human Centric Applications With Assured Security, Privacy, and Trust: From 5G to 6G." 2021. Accessed: Aug. 8, 2022. [Online]. Available: https://5g-ppp.eu/dedicat-6g/
- [51] "Resilient Interactive Applications Through Hyper Diversity in Energy Efficient RadioWeaves Technology." 2021. Accessed: Aug. 8, 2022. [Online]. Available: https://reindeer-project.eu/
- [52] U. K. Ganesan, E. Bjornson, and E. G. Larsson, "RadioWeaves for extreme spatial multiplexing in indoor environments," in *Proc. 54th Asilomar Conf. Signals Syst. Comput.*, 2020, pp. 1007–1011.
- [53] R. Sarvendranath, U. K. Ganesan, Z. H. Shaik, and E. G. Larsson. "Physical Layer Abstraction Model for RadioWeaves." 2022. [Online]. Available: https://arxiv.org/abs/2206.13924
- [54] AI4Green. "AI4Green Project." 2019. Accessed: Jan. 25, 2023. [Online]. Available: https://ai4green.celfinet.com/
- [55] M. Masoudi, M. G. Khafagy, E. Soroush, D. Giacomelli, S. Morosi, and C. Cavdar, "Reinforcement learning for traffic-adaptive sleep mode management in 5G networks," in *Proc. IEEE Int. Symp. Pers. Indoor Mobile Radio Commun. (PIMRC)*, 2020, pp. 1–6.

- [56] C. Amaro, T. Saraiva, D. Duarte, P. Vieira, M. P. Queluz, and A. Rodrigues, "Developing a new simulation and visualization platform for researching aspects of mobile network performance," in *Proc. Int. Symp. Wireless Pers. Multimedia Commun. (WPMC)*, 2021, p. 6.
- [57] Mobile Net Zero: State of the Industry on Climate Action 2022, GSMA, London, U.K., 2022.
- [58] Deutsche Telekom, Orange, Telefónica, TIM and Vodafone. "Open Ran Technical Priorities Focus on Energy Efficiency." 2022. Accessed: Jan. 25, 2023. [Online]. Available: https:// telecominfraproject.com/OpenRAN-MoU-Group/
- [59] Next G Alliance. "Green G Working Group." 2021. Accessed: Jan. 26, 2023. [Online]. Available: https://www.nextgalliance.org/ working_group/green-g/
- [60] Technical Specification Group Services and System Aspects; Management and Orchestration; Energy Efficiency of 5G, V18.0.0, 3GPP Standard TS 28.310, 2022.
- [61] Going Green: Benchmarking the Energy Efficiency of Mobile, GSMA, London, U.K., 2021.
- [62] RF Editorial Team. "What Is a Remote Radio Head (RRH)?" 2019. Accessed: Jan. 19, 2023. [Online]. Available: https://www. everythingrf.com/community/what-is-a-remote-radio-head
- [63] M. KANEKO, "Active Antenna," in Proc. IEEE Int. Conf. Robot. Autom. (ICRA), 1994, pp. 2665–2671.
- [64] J. Lorincz, T. Garma, and G. Petrovic, "Measurements and modelling of base station power consumption under real traffic loads," *Sensors*, vol. 12, no. 4, pp. 4281–4310, 2012.
- [65] A. Ayang, P.-S. Ngohe-Ekam, B. Videme, and J. Temga, "Power consumption: Base stations of telecommunication in Sahel zone of cameroon: Typology based on the power consumption—Model and energy savings," *J. Energy*, vol. 2016, Jul. 2016, Art. no. 3161060, doi: 10.1155/2016/3161060.
- [66] L. M. P. Larsen, H. L. Christiansen, S. Ruepp, and M. S. Berger, "Deployment guidelines for cloud-RAN in future mobile networks," in *Proc. IEEE CloudNet*, 2022, pp. 141–149.
- [67] Red Hat. "VNF and CNF, What's the Difference?" 2022. Accessed: Aug. 17, 2022. [Online]. Available: https://www.redhat.com/en/ topics/cloud-native-apps/vnf-and-cnf-whats-the-difference
- [68] A. Mughees, M. Tahir, M. A. Sheikh, and A. Ahad, "Towards energy efficient 5G networks using machine learning: Taxonomy, research challenges, and future research directions," *IEEE Access*, vol. 8, pp. 187498–187522, 2020.
- [69] Zain Group, stc, e&, Mobily, du, Batelco and Omantel. "Open RAN for Brownfield Operators Challenges and Opportunities." 2022. [Online]. Available: https://telecominfraproject.com/press-coverage
- [70] P. Ledl, What Is the Power-Saving Potential Based on the vRAN Deployment in DT Bonn Lab? Tecknexus, Oro Valley, AZ, USA, 2022.
- [71] S. Friis, L. M. P. Larsen, and S. Ruepp, "Strategies for minimization of energy consumption in data Centers," in *Proc. IARIA 22nd Int. Conf. Netw. ICN*, 2023, pp. 2–9.
- [72] A. N. Al-Quzweeni, A. Q. Lawey, T. E. Elgorashi, and J. M. Elmirghani, "Optimized energy aware 5G network function virtualization," *IEEE Access*, vol. 7, pp. 44939–44958, 2019.
- [73] L. M. P. Larsen, M. S. Berger, and H. L. Christiansen, "Energyaware design considerations for Ethernet-based 5G mobile Fronthaul networks," in *Proc. 4th Int. Conf. Green Commun. Comput. Technol.*, 2019, pp. 1–8.
- [74] L. M. P. Larsen, M. S. Berger, and H. L. Christiansen, "Energyaware technology comparisons for 5G mobile Fronthaul networks," *Int. J. Adv. Netw. Services*, vol. 13, nos. 1–2, pp. 21–32, 2020.
- [75] W. Xia, J. Zhang, T. Q. Quek, S. Jin, and H. Zhu, "Energy-efficient task scheduling and resource allocation in downlink C-RAN," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, 2018, pp. 1–6.
- [76] A. Younis, T. X. Tran, and D. Pompili, "Fronthaul-aware resource allocation for energy efficiency maximization in C-RANs," in *Proc.* 15th IEEE Int. Conf. Auton. Comput. (ICAC), 2018, pp. 1–6.
- [77] O-RAN.WG9.XTRP-REQ-v01.00: Xhaul Transport Requirements, O-RAN Alliance, Alfter, Germany, 2021.
- [78] NGMN Overview on 5G RAN Functional Decomposition, NGMN Alliance, Frankfurt, Germany, 2018.
- [79] IEEE Standard for Packet-Based Fronthaul Transport Network, IEEE Std 1914.1-2019, 2019.

- [80] "Study on new radio access technology: Radio access architecture and interfaces, v14.0.0," 3GPP, Sophia Antipolis, France, Rep. TR 38.801, 2017.
- [81] L. M. P. Larsen, M. S. Berger, and H. L. Christiansen, "Fronthaul for cloud-RAN enabling network slicing in 5G mobile networks," *Wireless Commun. Mobile Comput.*, vol. 2018, Aug. 2018, Art. no. 4860212.
- [82] M. R. Aktar and M. S. Anower, "Improvement of spectral and energy efficiency by coordinated transmission in C-RAN," in *Proc. 6th Int. Conf. Comput. Commun. Chem. Mater. Electron. Eng. (IC4ME2)*, 2021, p. 4.
- [83] F. Z. Morais, C. A. da Costa, A. M. Alberti, C. B. Both, and R. da Rosa Righi, "When SDN meets C-RAN: A survey exploring multi-point coordination, interference, and performance," *J. Netw. Comput. Appl.*, vol. 162, Jul. 2020, Art. no. 102655.
- [84] M. Zhu, J. Gu, X. Zeng, C. Yan, and P. Gu, "Delay-aware energy-saving strategies for BBU pool in C-RAN: Modeling and optimization," *IEEE Access*, vol. 9, pp. 63257–63266, 2021.
- [85] J. Na, J. Koh, S. Park, and J. Kang, "Energy efficiency enhancement on cloud and edge processing by dynamic RRH selection," in *Proc. 15th IEEE Annu. Consum. Commun. Netw. Conf. (CCNC)*, 2018, pp. 1–6.
- [86] M. R. Aktar and M. S. Anower, "Improvement of energy efficiency by dynamic load consolidation in C-RAN," *Int. J. Commun. Syst.*, vol. 35, no. 6, 2022, Art. no. e5087.
- [87] M. Alemam, A. A. El-Sherif, and T. Elbatt, "Energy efficiency optimization through RRHs ON/OFF switching technique in C-RAN," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, 2019, pp. 1–8.
- [88] R. S. Alhumaima and H. S. Al-Raweshidy, "Modelling the power consumption and trade-offs of virtualised cloud radio access networks," *IET Commun.*, vol. 11, no. 7, pp. 1158–1164, 2017.
- [89] V. Q. Rodriguez and F. Guillemin, "Cloud-RAN modeling based on parallel processing," *IEEE J. Sel. Areas Commun.*, vol. 36, no. 3, pp. 457–468, Mar. 2018.
- [90] B. J. Sahu, S. Dash, N. Saxena, and A. Roy, "Energy-efficient BBU allocation for green C-RAN," *IEEE Commun. Lett.*, vol. 21, no. 7, pp. 1637–1640, Jul. 2017.
- [91] J. Francis and G. Fettweis, "Energy efficiency maximization in massive MIMO-aided, Fronthaul-constrained C-RAN," in *Proc. IEEE Int. Symp. Pers. Indoor Mobile Radio Commun. (PIMRC)*, 2019, pp. 1–6.
- [92] Ericsson. "Ericsson 5G Portfolio Update Puts Energy Efficiency Center Stage." 2022. Accessed: Jul. 11, 2022. [Online]. Available: https://www.ericsson.com/en/press-releases/2022/2/ericsson-5g-portf olio-update-puts-energy-efficiency-center-stage
- [93] J. Hultell and M. Begley. "A Holistic Approach to Address RAN Energy Efficiency." 2021. Accessed: Aug. 15, 2022. [Online]. Available: https://www.ericsson.com/en/blog/2021/12/aholistic-approach-to-address-ran-energy-efficiency
- [94] Y. Lin, L. Shao, Z. Zhu, Q. Wang, and R. K. Sabhikhi, "Wireless network cloud: Architecture and system requirements," Int. Bus. Mach., Armonk, NY, USA, Rep. TR-2010, 2010.
- [95] C-RAN the Road Towards Green RAN, China Mobile Res. Inst., Beijing, China, 2011.
- [96] R. Bassoli, F. Granelli, S. T. Arzo, and M. Di Renzo, "Toward 5G cloud radio access network: An energy and latency perspective," *Trans. Emerg. Telecommun. Technol.*, vol. 32, no. 1, 2021, Art. no. e3669.
- [97] W. Jiang, B. Han, M. A. Habibi, and H. D. Schotten, "The road towards 6G: A comprehensive survey," *IEEE Open J. Commun. Soc.*, vol. 2, pp. 334–366, 2021.
- [98] Vodafone. "Vodafone, Wind River, Intel, Keysight Technologies and Radisys Test 'Green' Open RAN Network." 2022. Accessed: Jul. 11, 2022. [Online]. Available: https://www.vodafone.com/news/ technology/vodafone-wind-river-intel-keysight-technologies-radisystest-green-open-ran
- [99] Open RAN: Mature and Ready for Deployment, Mavenir, Richardson, TX, USA, 2021.
- [100] T. Pamuklu, S. Mollahasani, and M. Erol-Kantarci, "Energy-efficient and delay-guaranteed joint resource allocation and du selection in O-RAN," in *Proc. IEEE 4th 5G World Forum (5GWF)*, 2021, pp. 99–104.

- [101] S. Mollahasani, T. Pamuklu, R. Wilson, and M. Erol-Kantarci, "Energy-aware dynamic DU selection and NF relocation in O-RAN using actor–critic learning," *Sensors*, vol. 22, no. 13, p. 5029, 2022.
- [102] A. K. Singh and K. K. Nguyen, "Joint selection of local trainers and resource allocation for federated learning in open RAN intelligent controllers," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, 2022, pp. 1874–1879.
- [103] B. Brik, K. Boutiba, and A. Ksentini, "Deep learning for B5G open radio access network: Evolution, survey, case studies, and challenges," *IEEE Open J. Commun. Soc.*, vol. 3, pp. 228–250, 2022.
- [104] A. M. Abdelbar, A. M. El-Tager, and H. S. El-Hennawy, "Radio frequency power amplifier for green communications based on 10W GaN HEMT," in *Proc. 35th Nat. Radio Sci. Conf. (NRSC)*, 2018, pp. 352–360.
- [105] S. Zhang, C. Xiang, S. Cao, S. Xu, and J. Zhu, "Dynamic carrier and power amplifier mapping for energy efficient multi-carrier wireless communications," in *Proc. IEEE Int. Conf. Commun.*, 2018, pp. 1–6.
- [106] O. Pliushch, Y. Kravchenko, O. Matviichuk-Yudina, A. Rybydajlo, O. Trush, and R. Mykolaichuk, "Computer model of radio frequency power amplifier," in *Proc. IEEE 3rd Int. Conf. Adv. Trends Inf. Theory* (ATIT), 2021, pp. 159–164.
- [107] J. Lorincz, A. Capone, and J. Wu, "Greener, energy-efficient and sustainable networks: State-of-the-art and new trends," *Sensors*, vol. 19, no. 22, p. 4864, 2019.
- [108] A. Salh et al., "Low computational complexity for optimizing energy efficiency in mm-Wave hybrid precoding system for 5G," *IEEE Access*, vol. 10, pp. 4714–4727, 2022.
- [109] Y.-H. Zhu, G. Callebaut, L. Van der Perre, and F. Rottenberg, "Array placement in distributed massive MIMO for power saving considering environment information," in *Proc. VTC-Fall*, 2021, p. 6.
- [110] Ericsson. "Breaking the Energy Curve." 2020. Accessed: Sep. 14, 2022. [Online]. Available: https://ericsson.com/ breaking-the-energycurve
- [111] R. Tan, Y. Shi, Y. Fan, W. Zhu, and T. Wu, "Energy saving technologies and best practices for 5G radio access network," *IEEE Access*, vol. 10, pp. 51747–51756, 2022.
- [112] F. E. Salem, T. Chahed, Z. Altman, and A. Gati, "Traffic-aware advanced sleep modes management in 5G networks," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, 2019, pp. 1–9.
- [113] E. Schiller, J. Ajayi, S. Weber, T. Braun, and B. Stiller, "Toward a live BBU container migration in wireless networks," *IEEE Open J. Commun. Soc.*, vol. 3, pp. 301–321, 2022.
- [114] B. Mao, F. Tang, Y. Kawamoto, and N. Kato, "AI models for green communications towards 6G," *IEEE Commun. Surveys Tuts.*, vol. 24, no. 1, pp. 210–247, 1st Quart., 2022.
- [115] Z. M. Liu, C. Zhang, and P. S. Yu, "Direction-of-arrival estimation based on deep neural networks with robustness to array imperfections," *IEEE Trans. Antennas Propag.*, vol. 66, no. 12, pp. 7315–7327, Dec. 2018.
- [116] J. Thrane, D. Zibar, and H. L. Christiansen, "Model-aided deep learning method for path loss prediction in mobile communication systems at 2.6 GHz," *IEEE Access*, vol. 8, pp. 7925–7936, 2020.
- [117] A. Thantharate, A. V. Tondwalkar, C. Beard, and A. Kwasinski, "ECO6G: Energy and cost analysis for network slicing deployment in beyond 5G networks," *Sensors*, vol. 22, no. 22, p. 8614, 2022.
- [118] Nokia. "AVA AI and Analytics." Accessed: Feb. 2, 2023. [Online]. Available: https://www.nokia.com/networks/ai-and-analytics/#whatis-AVA-telecom-analytics
- [119] Huawei. "Intelligent Energy Network." Accessed: Feb. 2, 2023. [Online]. Available: https://digitalpower.huawei.com/en/site-powerfacility/scene_solution/intelligent/intelligent.html
- [120] ZTE. "PowerPilot." Accessed: Feb. 2, 2023. [Online]. Available: https://www.zte.com.cn/global/solutions_latest/wireless_access/green2.html
- [121] Samsung. "Discovering New Insights With Samsung VISTA." Accessed: Feb. 2, 2023. [Online]. Available: https://www.samsung. com/global/business/networks/products/network-automation/vista/
- [122] Ericsson. "AI-Powered Energy Management." Accessed: Feb. 2, 2023. [Online]. Available: https://www.ericsson.com/ en/managed-services/energy-infrastructure-operations
- [123] Coursera. "Deep Learning vs. Machine Learning: Beginner's Guide." 2022. Accessed: Feb. 5, 2023. [Online]. Available: https:// www.coursera.org/articles/ai-vs-deep-learning-vs-machine-learningbeginners-guide

- [124] A. Zappone, M. Di Renzo, M. Debbah, T. T. Lam, and X. Qian, "Model-aided wireless artificial intelligence: Embedding expert knowledge in deep neural networks for wireless system optimization," *IEEE Veh. Technol. Mag.*, vol. 14, no. 3, pp. 60–69, Sep. 2019.
- [125] D. Korpi. "6G Networks Will Be Energy Efficient From the Get-Go Thanks to AI/ML." 2021. Accessed: Aug. 17, 2022. [Online]. Available: https://www.bell-labs.com/institute/blog/6g-networks-willbe-energy-efficient-from-the-get-go-thanks-to-aiml/#gref
- [126] S. Sharma and W. Yoon, "Multiobjective reinforcement learning based energy consumption in C-RAN enabled massive MIMO," *Radioengineering*, vol. 31, no. 1, pp. 155–163, 2022.
- [127] S. K. G. Peesapati, M. Olsson, and S. Andersson, "A multi-strategy multi-objective hierarchical approach for energy management in 5G networks," in *Proc. Globecom IEEE Global Commun. Conf.*, 2022, pp. 4842–4847.
- [128] S. K. G. Peesapati, M. Olsson, M. Masoudi, S. Andersson, and C. Cavdar, "Q-learning based radio resource adaptation for improved energy performance of 5G base stations," in *Proc. IEEE Int. Symp. Pers. Indoor Mobile Radio Commun. (PIMRC)*, 2021, pp. 979–984.
- [129] A. Foivos, B. Christos, and K. Vasileios, "A machine learning mechanism for improved energy consumption in 5G small cells," in *Proc. IEEE Symp. Comput. Commun.*, 2022, p. 6.
- [130] G. Du, L. Wang, Q. Liao, and H. Hu, "Deep neural network based cell sleeping control and beamforming optimization in cloud-RAN," in *Proc. IEEE Veh. Technol. Conf.*, 2019, p. 6.
- [131] Y. Xing, Y. Qian, W. Shi, X. Liu, T. Zhao, and C. Tapparello, "Deep learning for Optimized multiuser OFDMA energy-efficient wireless transmission," in *Proc. IEEE 11th Annu. Comput. Commun. Workshop Conf. (CCWC)*, 2021, pp. 578–584.
- [132] F. B. Mismar, A. Alammouri, A. Alkhateeb, J. G. Andrews, and B. L. Evans, "Deep learning predictive band switching in wireless networks," *IEEE Trans. Wireless Commun.*, vol. 20, no. 1, pp. 96–109, Jan. 2021.
- [133] J. S. Pujol-Roigl, S. Wu, Y. Wang, M. Choi, and I. Park, "Deep reinforcement learning for cell on/off energy saving on wireless networks," in *Proc. IEEE Global Commun. Conf. (Globecom)*, 2021, p. 7.
- [134] H. S. Lee, D. Y. Kim, and J. W. Lee, "Radio and energy resource management in renewable energy-powered wireless networks with deep reinforcement learning," *IEEE Trans. Wireless Commun.*, vol. 21, no. 7, pp. 5435–5449, Jul. 2022.
- [135] G. Vallero, D. Renga, M. Meo, and M. A. Marsan, "Greener RAN operation through machine learning," *IEEE Trans. Netw. Service Manag.*, vol. 16, no. 3, pp. 896–908, Sep. 2019.
- [136] M. Liyanage et al., "A survey on zero touch network and service management (ZSM) for 5G and beyond networks," J. Netw. Comput. Appl., vol. 203, Jul. 2022, Art. no. 103362.
- [137] F. Rezazadeh, H. Chergui, L. Alonso, and C. Verikoukis, "Continuous multi-objective zero-touch network slicing via twin delayed DDPG and OpenAI gym," in *Proc. IEEE Global Commun. Conf. (Globecom)*, 2020, pp. 1–6.
- [138] L. Bonati et al., "CellOS: Zero-touch softwarized open cellular networks," *Comput. Netw.*, vol. 180, Oct. 2020, Art. no. 107380.
- [139] F. Rezazadeh, H. Chergui, and C. Verikoukis. "Zero-Touch Continuous Network Slicing Control Via Scalable Actor–Critic Learning." 2021. [Online]. Available: https://arxiv.org/pdf/2101. 06654.pdf
- [140] K. Weinberger, A. A. Ahmad, A. Sezgin, and A. Zappone, "Synergistic benefits in IRS- and RS-enabled C-RAN with energyefficient clustering," *IEEE Trans. Wireless Commun.*, vol. 21, no. 10, pp. 8459–8475, Oct. 2022.
- [141] Y. Mao, O. Dizdar, B. Clerckx, R. Schober, P. Popovski, and H. V. Poor. "Rate-Splitting Multiple Access: Fundamentals, Survey, and Future Research Trends." 2022. [Online]. Available: https://arxiv. org/abs/2201.03192
- [142] A. A. Ahmad, B. Matthiesen, A. Sezgin, and E. Jorswieck, "Energy efficiency in C-RAN using rate splitting and common message decoding," in *Proc. IEEE Int. Conf. Commun. Workshops (ICC Workshops)*, 2020, pp. 1–6.
- [143] A. A. Ahmad, Y. Mao, A. Sezgin, and B. Clerckx, "Rate splitting multiple access in C-RAN: A scalable and robust design," *IEEE Trans. Commun.*, vol. 69, no. 9, pp. 5727–5743, Sep. 2021.

- [144] J. Zhang, J. Zhang, Y. Zhou, H. Ji, J. Sun, and N. Al-Dhahir, "Energy and spectral efficiency tradeoff via rate splitting and common beamforming coordination in multicell networks," *IEEE Trans. Commun.*, vol. 68, no. 12, pp. 7719–7731, Dec. 2020.
- [145] M. Z. Chowdhury, M. Shahjalal, S. Ahmed, and Y. M. Jang, "6G wireless communication systems: Applications, requirements, technologies, challenges, and research directions," *IEEE Open J. Commun. Soc.*, vol. 1, pp. 957–975, 2020.
- [146] M. Shehab, B. S. Ciftler, T. Khattab, M. M. Abdallah, and D. Trinchero, "Deep reinforcement learning powered IRS-assisted Downlink NOMA," *IEEE Open J. Commun. Soc.*, vol. 3, pp. 729–739, 2022.
- [147] G. C. Trichopoulos et al., "Design and evaluation of reconfigurable intelligent surfaces in real-world environment," *IEEE Open* J. Commun. Soc., vol. 3, pp. 462–474, 2022.
- [148] C. D. Alwis et al., "Survey on 6G frontiers: Trends, applications, requirements, technologies and future research," *IEEE Open J. Commun. Soc.*, vol. 2, pp. 836–886, 2021.
- [149] J. Chen, Y. Xie, X. Mu, J. Jia, Y. Liu, and X. Wang, "Energy efficient resource allocation for IRS assisted CoMP systems," *IEEE Trans. Wireless Commun.*, vol. 21, no. 7, pp. 5688–5702, Jun. 2022.
- [150] Z. Mohamed and S. Aïssa, "Resource allocation for energy-efficient cellular communications via aerial IRS," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, 2021, p. 6.
- [151] J. Liang, L. Li, and C. Zhao, "A transfer learning approach for compressed sensing in 6G-IoT," *IEEE Internet Things J.*, vol. 8, no. 20, pp. 15276–15283, Oct. 2021.
- [152] M. Leinonen and M. Codreanu, "Quantized compressed sensing via deep neural networks," in *Proc. 2nd 6G Wireless Summit Gain Edge* 6G Era 6G Summit, 2020, p. 6.
- [153] V. Upadhyaya and D. M. Salim, "Compressive sensing: Methods, techniques, and applications," in *Proc. IOP Conf. Mater. Sci. Eng.*, vol. 1099, Mar. 2021, Art. no. 012012. doi: 10.1088/1757-899X/1099/1/012012.
- [154] H. Sun, F. Han, S. Zhao, and H. Deng, "Optimal energy-harvesting design for AF and DF two-way relay beamforming in 6G," *EURASIP* J. Wireless Commun. Netw., vol. 2022, no. 1, p. 73, 2022.
- [155] J. M. Kang, I. M. Kim, and D. I. Kim, "Joint optimal mode switching and power adaptation for nonlinear energy harvesting SWIPT system over fading channel," *IEEE Trans. Commun.*, vol. 66, no. 4, pp. 1817–1832, Apr. 2018.
- [156] C. Chen, X. Li, H. Ji, and H. Zhang, "Energy-efficient mobile edge computing system based on full-duplex energy harvesting relay network," in *Proc. IEEE Global Commun. Conf. (Globecom)*, 2020, p. 6.
- [157] J. Li, K. Zhao, X. Li, X. Ding, and W. Shi, "Research on interference energy harvesting based on SWIPT relay system," in *Proc. Soc. Inf. Telecommun. Eng.*, vol. 230, 2018, pp. 430–439.
- [158] N. Piovesan, A. F. Gambin, M. Miozzo, M. Rossi, and P. Dini, "Energy sustainable paradigms and methods for future mobile networks: A survey," *Comput. Commun.*, vol. 119, pp. 101–117, Apr. 2018, [Online]. Available: https://doi.org/10.10%2Fj.comcom. 2018.01.005
- [159] A. P. Couto Da Silva, D. Renga, M. Meo, and M. Ajmone Marsan, "On the use of small solar panels and small batteries to reduce the RAN carbon footprint," in *Proc. Mediterr. Commun. Comput. Netw. Conf. (Medconnet)*, 2020, p. 6.
- [160] M. Antal et al., "DC thermal energy flexibility model for waste heat reuse in nearby Neighborhoods," in *Proc. 8th Int. Conf. Future Energy Syst. (e-Energy)*, 2017, pp. 278–283. [Online]. Available: https://doi-org.proxy.findit.cvt.dk/10.1145/3077839.3084024
- [161] S. Zimmermann, M. K. Tiwari, I. Meijer, S. Paredes, B. Michel, and D. Poulikakos, "Hot water cooled electronics: Exergy analysis and waste heat reuse feasibility," *Int. J. Heat Mass Transfer*, vol. 55, no. 23, pp. 6391–6399, 2012. [Online]. Available: https:// www.sciencedirect.com/science/article/pii/S0017931012004553
- [162] Y. Zhen and Y. Li, "Redox flow battery," *Stud. Surf. Sci. Catalysis*, vol. 179, pp. 385–413, Jan. 2019.
- [163] A. Bousia and A. Daskalopulu, "Double auction offloading: Challenges and opportunities for energy efficient telecommunication networks," in *Proc. ACM Int. Conf.*, 2021, pp. 92–97.

- [164] X. Marjou, T. Le Gleau, V. Messie, B. Radier, T. Lemlouma, and G. Fromentoux, "Evaluating inter-operator cooperation scenarios to save radio access network energy," in *Proc. 1st Int. Conf. 6G Netw.* (6GNET), 2022, p. 5.
- [165] Network Sharing; Architecture and Functional Description, V17.0.0, 3GPP Standard TS 23.251, 2022.
- [166] Network Sharing; Concepts and Requirements, V17.4.0, 3GPP Standard TS 32.130, 2022.
- [167] X. Zhao, C. Hu, and Z. Li, "Multi-operator radio access network sharing for 5G SA network design and laboratory test," in *Proc. Int. Wireless Commun. Mobile Comput. (IWCMC)*, 2021, pp. 187–193.
- [168] Y. Zhao, P. Li, and C. Hu, "Verification and evaluation on 5G multi-operator core network," in *Proc. Int. Wireless Commun. Mobile Comput. (IWCMC)*, 2021, pp. 498–502.
- [169] M. Kassis, S. Costanzo, and M. Yassin, "Flexible multi-operator RAN sharing: Experimentation and validation using open source 4G/5G prototype," in *Proc. Joint Eur. Conf. Netw. Commun. 6G Summit* eUCNC/6G Summit, 2021, pp. 205–210.
- [170] J. Olesen. "TT-netværket, Telenor og telia, sammenlægger nu 2Gnettet." 2014. Accessed: Dec. 8, 2022. [Online]. Available: https:// meremobil.dk/2014/09/tt-netvaerket-sammenlaegger-2g/
- [171] D. Leza. "Mobile infrastructure sharing: Trends in Latin America." 2014. [Online]. Available: https://www.itu.int/en/ITU-D/Regulatory-Market/Documents/CostaRica/Presentations/Session8_Daniel% 20Leza%20-%20Mobile%20Infrastructure%20Sharing%20-%2012% 20March%202014.pdf
- [172] X. Zhao, J. Chen, P. Li, Z. Li, C. Hu, and W. Xie, "5G NSA radio access network sharing for mobile operators: Design, realization and field trial," in *Proc. Int. Conf. Commun. Technol. Process. (ICCT)*, 2020, pp. 454–461.
- [173] A. Weissberger. "Nokia Deploys Shared 5G RAN (MORAN) With SoftBank and KDDI in Japan." 2021. Accessed: Dec. 8, 2022. [Online]. Available: https://techblog.comsoc.org/2021/10/14/nokiadeploys-shared-5g-ran-moran-with-softbank-and-kddi-in-japan/
- [174] P. Kokkinos, P. Soumplis, A. Kretsis, and E. M. Varvarigos, "Network slicing in 5G infrastructures from the edge to the core," in *Proc. 12th Int. Symp. Commun. Syst. Netw. Digit. Signal Process. (CSNDSP)*, 2020, p. 6.
- [175] X. Li, R. Ni, J. Chen, Y. Lyu, Z. Rong, and R. Du, "End-to-end network slicing in radio access network, transport network and core network domains," *IEEE Access*, vol. 8, pp. 29525–29537, 2020.
- [176] M. Chahbar, G. Diaz, A. Dandoush, C. Cerin, and K. Ghoumid, "A comprehensive survey on the E2E 5G network slicing model," *IEEE Trans. Netw. Service Manag.*, vol. 18, no. 1, pp. 49–62, Mar. 2021.
- [177] C. M. Mohammed and S. K. Shaikhah, "A survey and analysis of architecture and models of network slicing in 5G," in *Proc. 8th Int. Eng. Conf. Sustain. Technol. Develop. (IEC)*, 2022, pp. 192–198.
- [178] M. Masoudi, O. T. Demir, J. Zander, and C. Cavdar, "Energy-optimal end-to-end network slicing in cloud-based architecture," *IEEE Open J. Commun. Soc.*, vol. 3, pp. 574–592, 2022.
- [179] C. Sexton, N. Marchetti, and L. A. Dasilva, "Customization and trade-offs in 5G RAN slicing," *IEEE Commun. Mag.*, vol. 57, no. 4, pp. 116–122, Feb. 2019.
- [180] C. Bektas, S. Bocker, F. Kurtz, and C. Wietfeld, "Reliable softwaredefined RAN network slicing for mission-critical 5G communication networks," in *Proc. IEEE Globecom Workshops (GC Wkshps)*, 2019, p. 6.
- [181] Y. Azimi, S. Yousefi, H. Kalbkhani, and T. Kunz, "Energy-efficient deep reinforcement learning assisted resource allocation for 5G-RAN slicing," *IEEE Trans. Veh. Technol.*, vol. 71, no. 1, pp. 856–871, Jan. 2022.
- [182] D. Ginthör, R. Guillaume, M. Schüngel, and H. D. Schotten, "5G RAN slicing for deterministic traffic," in *Proc. IEEE Wireless Commun. Netw. Conf. (WCNC)*, 2021, p. 6.
- [183] L. Feng, Y. Zi, W. Li, F. Zhou, P. Yu, and M. Kadoch, "Dynamic resource allocation with RAN slicing and scheduling for uRLLC and eMBB hybrid services," *IEEE Access*, vol. 8, pp. 34538–34551, 2020.
- [184] N. J. Gomes et al., "Boosting 5G through Ethernet: How evolved Fronthaul can take next-generation mobile to the next level," *IEEE Veh. Technol. Mag.*, vol. 13, no. 1, pp. 74–84, Apr. 2018.
- [185] IEEE Standard for Local and Metropolitan Area Network-Bridges and Bridged Networks, IEEE Standard 802.1Q-2018, 2018.

- [186] D. Liang, R. Gu, Q. Guo, and Y. Ji, "Demonstration of multi-vendor multi-standard pon networks for network slicing in 5G-oriented mobile network," in *Proc. Asia Commun. Photon. Conf. (ACP)*, 2017, p. 5.
- [187] M. Ruffini, A. Ahmad, S. Zeb, N. Afraz, and F. Slyne, "Virtual DBA: Virtualizing passive optical networks to enable multi-service operation in true multi-tenant environments," *J. Opt. Commun. Netw.*, vol. 12, no. 4, pp. B63–B73, 2020.
- [188] M. Ruffini and F. Slyne, "Moving the network to the cloud: The cloud central office revolution and its implications for the optical layer," *J. Lightw. Technol.*, vol. 37, no. 7, pp. 1706–1716, Apr. 1, 2019.
- [189] F. Slyne, B. Cornaglia, M. Boselli, and M. Ruffini, "Single-stage scheduler for accurate QoS delivery in virtualised multi-tenant passive optical networks," in *Proc. IET Conf.*, 2019, p. 4.
- [190] C. H. T. Arteaga, A. Ordoñez, and O. M. C. Rendon, "Scalability and performance analysis in 5G core network slicing," *IEEE Access*, vol. 8, pp. 142086–142100, 2020.
- [191] K. Noroozi, M. Karimzadeh-Farshbafan, and V. Shah-Mansouri, "Service admission control for 5G mobile networks with RAN and core slicing," in *Proc. IEEE Global Commun. Conf. (Globecom)*, 2019, pp. 1–6.
- [192] S. Vittal, M. K. Singh, and A. A. Franklin, "Adaptive network slicing with multi-site deployment in 5G core networks," in *Proc. Conf. Netw. Softw. Bridging Gap AI Netw. Softw. Netsoft*, 2020, pp. 227–231.
- [193] J. O'Halloran, "Samsung serves DISH for 5G open RAN roll-out." 2022. Accessed: Oct. 21, 2022. [Online]. Available: https://www. computerweekly.com/news/252516709/Samsung-serves-DISH-for-5G-Open-RAN-roll-out
- [194] Rakuten. "Mythbusting Open RAN—Separating Fact From Fiction on Open Radio Access Networks." 2021. Accessed: Oct. 21, 2022.
 [Online]. Available: https://rakuten.today/blog/mythbusting-open-ran. html
- [195] I. Morris. "Cloud RAN Still a Hard Sell Despite Vodafone and Orange Interest." 2021. Accessed: Oct. 21, 2022. [Online]. Available: https://www.lightreading.com/open-ran/cloud-ran-stillhard-sell-despite-vodafone-and-orange-interest/d/d-id/772644
- [196] Rakuten Symphony. "1&1 and Rakuten Team Up to Build Europe's First Fully Virtualized Nationwide Mobile Network Based on Open RAN Technology." 2022. Accessed: Oct. 21, 2022. [Online]. Available: https://symphony.rakuten.com/blog/1and1-rakuten-teamto-build-europes-first-fully-virtualized-nationwide-mobile-networkbased-on-open-ran-tech



LINE M. P. LARSEN (Member, IEEE) received the B.Sc. (Eng.) degree in IT and communications technology and the M.Sc. (Eng.) degree in telecommunication from the Technical University of Denmark (DTU) in 2015 and 2017, respectively, and the Ph.D. degree from the Department of Photonics Engineering, DTU in 2022.

Since 2022, she has been with Danish Network Operator TDC Net as an Industrial Postdoctoral, on a research project in collaboration with the Department of Electrical and Photonics

Engineering, DTU. She has previously worked in the area of Cloud-RAN mobile networks focusing on fronthaul, midhaul, and crosshaul/xhaul connections and functional splits. Her current research interests includes energy efficient and green mobile networks and future network architectures, including open RAN and the road toward 6G.

Dr. Larsen received the Best Paper Award at the International Conference on Green Communications, Computing and Technologies, GREEN 2019.



HENRIK L. CHRISTIANSEN (Member, IEEE) received the M.Sc. degree in electrical engineering and Ph.D. degree in telecommunications from the Technical University of Denmark, where he has been an Associate Professor of Mobile Communication focusing on mobile network architectures and evolution. He has also spent several years in the telecom industry and is currently a Senior Mobile RAN Architect with TDC NET, which is the largest mobile operator in Denmark.



SARAH RUEPP (Member, IEEE) received the B.Sc. degree in electronic and computer engineering in 2002, the M.Sc. degree in telecommunication in 2004, and the Ph.D. degree in reliable communication networks in 2008. She is an Associate Professor with the DTU Electro, Technical University of Denmark. Her research interests include topics related to mission critical communication systems, network reliability and survivability, Internet of Things, and 5G and beyond. She also serves as the head of study of the

DTU M.Sc. Programme Communication Technologies and System Design.



MICHAEL S. BERGER (Member, IEEE) was born in 1972. He received the M.Sc. degree in electrical engineering and the Ph.D. degree from the Technical University of Denmark (DTU) in 1998 and 2004, respectively.

He is currently an Associate Professor with the university within the area of network engineering and network node design. He has been participating in several projects with relation to energy efficient networking. He was a Project Leader on the national project Ethernet for RAN, where TSN

was explored in the Fronthaul Network. Furthermore, he coordinated the participation from DTU in a Eurostars Project on efficient mobile networks (Fronthaul for C-RAN) and participated in the FP7 COSIGN project about greening the data center networks by integrating optics and software defined networking. He is currently responsible for the Department participation in a Nordic University HUB project on Industrial IoT, Fog computing and TSN and he will become task leader for test and demonstrations in a new ambitious national project on Green Communications.